Stockholm Institute of Transition Economics

WORKING PAPER

January 2015 (updated November 2016)

No. 31

Preparing for Genocide: Quasi-Experimental Evidence from Rwanda

Evelina Bonnier, Jonas Poulsen, Thorsten Rogall and Miri Stryjan



Working papers from Stockholm Institute of Transition Economics (SITE) are preliminary by nature, and are circulated to promote discussion and critical comment. The views expressed here are the authors' own and not necessarily those of the Institute or any other organization or institution.

Stockholm Institute of Transition Economics (SITE) · Stockholm School of Economics · Box 6501 · SE-113 83 Stockholm · Sweden

Preparing for Genocide: Quasi-Experimental Evidence from Rwanda*

Evelina Bonnier[†] Jonas Poulsen[‡] Thorsten Rogall[§] Miri Stryjan[¶]

Abstract

This paper provides evidence for how an authoritarian state can mobilize civilians to participate in mass violence through a top-down policy. We analyze a Rwandan mandatory community program that required citizens to participate in community work and political meetings every Saturday in the years before the 1994 genocide. We exploit cross-sectional variation in meeting intensity induced by exogenous weather fluctuations, and find that a one standard-deviation increase in the number of rainy Saturdays resulted in a 16 percent lower civilian participation rate in genocide violence. The natural placebo test – rainfall on all other weekdays - yields no statistically significant results. Moreover, the result is entirely driven by areas under the control of pro-Hutu parties, and we find evidence that the political elites used the program beyond simple propaganda, bringing civilians together and practicing mobilization. We also present suggestive evidence that in areas with opposition parties in power, the effects turn positive, implying that the meetings there were used to overcome hatred. Our robust findings shed light on the potentially detrimental role of government-ordered community meetings. Its importance derives, at the very least, from the resurgence of similar practices in sub-Saharan Africa.

JEL classification: D74, N47

^{*}We thank Eli Berman, Benjamin Crost, Tom Cunningham, Gordon Dahl, Melissa Dell, Willa Friedman, Matt Gentzkow, Jonas Hjort, Chang-Tai Hsieh, Saumitra Jha, Juanna Joensen, Magnus Johannesson, Erik Lindqvist, Eva Mörk, Suresh Naidu, Nathan Nunn, Torsten Persson, Cristian Pop-Eleches, Marit Rehavi, Jacob Shapiro, David Strömberg, Jakob Svensson, Francesco Trebbi, Erik Verhoogen, and Miguel Urquiola, as well as seminar participants at Harvard, UCSD, Columbia, ESOC, ASWEDE, SSE, IIES, NEUDC, and the Annual Bank Conference on Africa for many helpful comments. Financial support from Handelsbanken's Research Foundations is gratefully acknowledged.

[†]Stockholm School of Economics, evelina.bonnier@hhs.se.

[‡]Harvard University, jonaspoulsen@fas.harvard.edu.

[§]University of British Columbia, thorsten.rogall@ubc.ca.

[¶]Ben-Gurion University of the Negev, miristry@bgu.ac.il.

1 Introduction

Large, disorganized groups that engage in protest, social unrest, or genocide first have to solve their collective action problem. To illustrate this, in several civil wars and conflicts, ordinary and seemingly unorganized civilians participate in violence. Are these people solely driven by deep-rooted hatred (bottom-up dynamics), or is the violence organized by the political leaders (top-down organization)? Civilian participation in violence often magnifies and escalates a given conflict with disastrous effects on the social fabric and the economy, not least the human suffering. It is therefore crucial to understand its causes. During the Rwandan Genocide in 1994, around 430,000 Hutu civilians joined the army and militiamen in killing an estimated 800,000 Tutsis and moderate Hutus in only 100 days. Anecdotal evidence for the Rwandan case points to a top-down preparation of the genocide: In the years before the genocide, the political elites abused the weekly-held community meetings called *Umuganda* (initially designed as mandatory work meetings to improve local infrastructure) to mobilize the civilian Hutu population against the ethnic Tutsi minority (Cook, 2004; Straus, 2006; Verwimp, 2013).

This paper provides the first empirical analysis of how important local, elite-led community meetings might have been in inducing the civilian population to participate in violence. Examining the possibly negative effect of these community meetings is of general importance: There is a widely held belief that community meetings foster social capital by providing arenas for people to meet, solve free-rider problems, and create public goods (Guiso, Sapienza and Zingalez, 2008; Grootaert and van Bastelaer, 2002; Knack and Keefer, 1997; Putnam, 2000). Consistently, many development agencies today focus on community-driven development projects in which deliberative forums and grass-root participation play a central role (see Mansuri and Rao (2012) for a recent overview). We investigate whether there is a "dark side" to these community meetings where social capital does not bridge the societal, ethnic divides but rather enforces bonding within groups, within the Hutu population in the Rwandan case. Understanding this process is even more important since *Umuganda* was formally re-introduced in Rwanda in 2008, and similar practices have been set up in Burundi, and are discussed

in the Democratic Republic of Congo and recently in Kenya (Daily Nation, 2016).

Furthermore, the Rwandan Genocide is part of a wider phenomenon of violent collective action such as civil conflict and social unrest. Understanding its mobilization dynamics is important for interpreting and handling other violent events involving civilian actors such as the 2007 post-election violence in Kenya. At the same time, the empirical literature on how groups solve their collective action problem is scarce.

Identifying the causal effect of these meetings on participation in genocide is difficult for two reasons. First, we lack data on the number of people participating in *Umuganda* or the number of meetings taking place in each area. Second, even if such data existed, our estimates would likely suffer from an omitted variable bias. On the one hand, area-specific unobservable characteristics that affect both genocide participation and *Umuganda* intensity, for instance local leader quality, could produce a spurious positive correlation between the two, thus biasing the estimate upwards. On the other hand, if *Umuganda* meetings were strategically used in areas where genocide participation was unobservably low, the estimate would be downward biased.

To overcome these data and endogeneity issues, we use exogenous rainfall variation to estimate the effect of *Umuganda* meetings on participation in civil conflict. The idea is simple: we expect the meetings to be less enjoyable when it rains and, furthermore, to be canceled altogether under heavy rains. Although we lack data to directly test this conjecture, it is supported by anecdotal evidence. Moreover, in the following analysis we present several tests that strongly suggest that our reduced-form effects are working via the *Umuganda* meetings. Since the community work took place on *Saturdays*, we can isolate the *Umuganda* effect from general rainfall effects (e.g. rainfall affecting income through agriculture) by only using the variation in Saturday rainfall while controlling for average daily rainfall. We use the number of Saturdays with heavy rainfall during the 3.5 year pre-genocide period (from October 1990, the outbreak of the civil war, to March 1994, the eve of the genocide) as our variable of interest.¹

Furthermore, we can provide a first placebo check by controlling for heavy rainfall

¹After the start of the civil war in October 1990, the tensions between Hutu and Tutsi intensified and the Hutu-dominated government became more aggressive towards the Tutsi minority, eventually culminating in the genocide. The increasing inter-ethnic hatred was for instance expressed in the government-funded journal *Kangura* (Melvern, 2004).

on all other six weekdays. This placebo test also allows us to rule out that other factors correlated with rainfall, such as geography, are driving the results as these factors should then be correlated with rainfall also on other weekdays. Nevertheless, we show that the results are robust to controlling for geographical factors such as sector ruggedness. To control for local characteristics, we include 142 commune fixed effects. Thus, we ensure that identification only stems from local variation in rainfall on Saturdays, which is arguably exogenous and should only affect genocide participation through its effect on *Umuganda* meeting intensity. There is, however, one major concern regarding the exclusion restriction: The effect we estimate might simply be a consequence of people getting together in general on Saturdays rather than the political leaders manipulating the population during the *Umuganda* meetings. We will argue in great detail why this concern is unwarranted.

We proxy for genocide violence by the number of people prosecuted in the Gacaca courts, normalized by sector Hutu population.² About 10,000 local Gacaca courts were set up all over the country to prosecute the crimes committed during the genocide. Importantly, these courts distinguished between civilian perpetrators and organized perpetrators such as members of militia gangs, the national army or the local police. Since organized perpetrators (mostly army and militiamen) moved around during the genocide, it is unclear whether the sector where they were exposed to pre-genocide *Umuganda* is the same as the locality where they appear in our prosecution data. To avoid this measurement error, we focus on civilian participation. Using prosecution instead of actual participation rates may introduce some bias. However, the Gacaca data is strongly correlated with other measures of genocide violence from various other sources and, using an alternative genocide measure from a different source, we present a number of additional tests to rule out that any systematic errors are biasing our results.

Our reduced-form results indicate a negative relationship between canceled *Umu-ganda* meetings (rainy Saturdays) and civilian participation in genocide violence: A one standard-deviation increase in the number of rainy Saturdays is associated with a 16 percent decrease in the civilian participation rate. Interestingly, this relationship is

²Sector is the second smallest administrative unit in Rwanda, with an average size of 14 square kilometers and 4,900 inhabitants.

entirely driven by sectors that are ruled by the pro-Hutu parties, and tracing the effects over time suggests that the meetings during the last 11 months before the genocide were the most important. This rules out long-term mechanisms, such as the infrastructure built during the meetings driving the results. Rather, we find evidence suggesting that the political elites used the program beyond simple propaganda, bringing people together and practicing mobilization. We also present suggestive empirical evidence that in areas with opposition parties in power, the effects turn positive, implying that the meetings there were used to create bonds between the two ethnicities.

Our results have important policy implications and are also relevant for other countries. Community meetings have recently been seen as a panacea for development by many NGOs and government agencies (Mansuri and Rao, 2012). Our results suggest that they can be captured and abused by local elites, thus calling for caution. In 2008, the Rwandan government reintroduced *Umuganda*. This might be worrisome since there is still tension between the Tutsi and Hutu population in the country. Furthermore, similar practices have been set up in Burundi and are being discussed in the Democratic Republic of Congo (DRC) and Kenya; all countries with a history of violent conflict along ethnic lines, which once more calls for caution when establishing an institution such as mandatory community meetings.

Our work contributes to the literature in several ways. Firstly, it advances the very scarce literature on the "dark side" of social capital (Satyanath, Voigtlaender and Voth, forthcoming), in contrast to several studies highlighting its positive effects (Guiso, Sapienza and Zingalez, 2008; Grootaert and van Bastelaer, 2002; Knack and Keefer, 1997). Secondly, it adds to a small empirical literature on how groups engaging in conflict, social unrest, or protest, solve their collective action problem. In line with Rogall (2014), we show that the Rwandan Genocide was carefully prepared by the political elites – a top-down approach. Thirdly, it complements the literature on the Rwandan Genocide (Friedman, 2013; Rogall, 2014; Verpoorten, 2012a-b; Verwimp, 2006; Yanagizawa-Drott, 2014) by providing novel evidence on its careful preparation.

Furthermore, Blattman and Miguel (2010) review the economics literature on conflict, vehemently calling for well-identified studies on the roots of individual participation in violent conflict. Recent studies on the determinants of conflict and participation in violence and killings consider government policy, income, foreign aid, and institutions (Dell, 2015; Mitra and Ray, 2014; Nunn and Qian, 2014; Dube and Vargas, 2013; Besley and Persson, 2011, respectively). Our study adds to this literature by providing novel evidence on the strong effects of local, elite-controlled community meetings on civilian participation in violence. The paper is also related to a literature in economics stressing the importance of political elites and their effects on institutions and conflict (Jones and Olken, 2009).

On the methodology side, our results add to the recent discussion on the effects of rainfall on conflict other than through the income channel (Iyer and Topalova, 2014; Rogall, 2014; Sarsons, 2015). Prominent studies that use various rainfall measures as instruments for income in Africa include Brückner and Ciccone (2010) and Miguel, Satyanath and Sergenti (2004).

The rest of the paper is organized as follows. Section 2 provides some background information on the Rwandan Genocide and *Umuganda*. Section 3 presents the data used for the analysis and Section 4 lays out our empirical strategy. Section 5 presents the main results and assesses their robustness, Section 6 discusses mechanisms and channels, and Section 7 concludes with possible policy implications.

2 Background

A History of Conflict Tensions between the Hutu and Tutsi populations have been present in Rwanda at least since colonial times. The origins of the two groups and the distinction between them is debated.³ What seems clear is that Belgian colonizers deepened the division between the two ethnic groups, and deliberately favored the Tutsi minority. Strong tensions rose between the two groups that culminated in the Rwandan revolution of 1959, where the Tutsi monarchy was replaced by a Hutu republic. During this period, many Tutsi civilians were killed; others fled Rwanda for neighboring

³The Tutsi minority (with a pre-genocide population share of around 10 percent) are said to have descended from Hamitic migrants from the north of Africa, and the Hutu majority from the Bantu group, who traditionally lived in Rwanda. However, others say that the two ethnicities do have a common ancestry.

countries such as Burundi, Tanzania and, in particular, Uganda. In the 1960s, episodes of political stability alternated with times of violence, but the underlying tensions never ceased.

In 1974 – paramount to the introduction of a modern version of *Umuganda* – Juvénal Habyarimana took power in Rwanda through a coup d'état. His subsequent rule was based on a pro-Hutu ideology ("Hutu power"), further discussed in the next section. In October 1990, the Rwandan Patriotic Front (RPF) invaded Rwanda from Uganda, starting the Rwandan civil war. The RPF was a Tutsi rebel army, who had emerged in exile, eager to replace the Hutu-led government. Fighting between the Hutu-led government and the Tutsi rebels continued until the Arusha Accords were signed in August 1993.⁴ A multi-party system was installed in the early phase of the peace talks, but this had little effect on reducing societal tension and conflict. On April 6 1994, an airplane with President Habyarimana on board was shot down over Kigali. Whether the Tutsis or Hutus are responsible for this attack remains unclear till this day, but quickly after the attack, extremists within the Hutu-dominated parties announced a new interim government and started a 100-day period of ethnic genocide throughout Rwanda. Around 800,000 people, mostly Tutsis and moderate Hutus lost their lives. The mass killings stopped in mid-July, when the RPF Tutsi rebels defeated the Rwandan Hutu army and the militia groups such as the Interahamwe.

A large number of Hutu civilians participated in the genocide violence, directed by the interim government (Dallaire, 2003). According to the Gacaca data that we use for this analysis, there were approximately 430,000 civilian perpetrators.⁵

Umuganda The practice of *Umuganda* dates back to pre-colonial times. During a day of community service, villagers would get together to build houses for the poor, or help each other out in the fields in times of economic hardship (Mukarubuga, 2006). Rather than being mandatory, *Umuganda* was initially considered a social obligation (Melvern, 2000). This changed during the colonial period, when the Belgian colonizers

⁴The essence of this treaty was a power-sharing government, including representatives from both sides of the conflict.

⁵For more information, see for example Straus (2006), Hatzfeld (2005), Dallaire (2003), Des Forges (1999), Gouveritch (1998), and Prunier (1995).

used *Umuganda* for organizing compulsory work. Consistently, the locally employed term for *Umuganda* in this period was *uburetwa*, or *forced labor* (IRDP, 2003). All men had to provide 60 days of communal work per year. Most of the manual labor was hereby carried out by members of the ethnic Hutu majority under the supervision of Tutsi chiefs (Pottier, 2006).

When *Umuganda* was re-introduced in 1974, the newly installed President Habyarimana, of the Hutu ethnic group, turned it into a political doctrine, which once again changed its meaning (Mamdani, 2001). Verwimp (2000, p. 344) cites Habyarimana:

"The doctrine of our movement [Movement for Development, MRND] is that Rwanda will only be developed by the sum of the efforts of its people. That is why it has judged the collective work for development a necessary obligation for all inhabitants of the country."

The program combined a practical motivation – achieving development objectives despite weak state finances – with a strong ideological element. Participation was again made compulsory through government coercion, and failure to participate usually involved paying a fine.⁶ The local leaders of the neighborhood who presided over a group of ten households were responsible for the weekly *Umugandas* and could decide who were to participate and demand fines from those failing to participate (Verwimp, 2000). The state chose the projects on which at least one adult male per family had to work on outdoor communal projects *every* Saturday morning (Uvin, 1998). According to a 1986 report, the main tasks consisted in anti-erosion measures and road maintenance work (Guichaoua, 1991).⁷ In the meetings held before or after the community work, local or higher officials disseminated information about the governing principles of the ruling party, The National Republican Movement for Democracy and Development (MRND), and messages from the government (Guichaoua, 1991).

Habyarimana's ideology stressed the importance of the cultivator as the true Rwan-

⁶In today's Rwanda, the fine for not participating in *Umuganda* is slightly less than \$10.

⁷Specifically, 56 percent of the work performed during *Umuganda* included various types of antierosion measures, such as terracing and digging ditches; 21 percent consisted of maintenance work of communal roads; 15 percent were construction of communal buildings while 3 percent were related to construction of water supply systems and another 3 percent were related to agriculture (Guichaoua, 1991).

dan (Straus, 2006). This view clearly embraced the Hutu population with their history as cultivators, as opposed to the Tutsi pastoralists. During the period leading up to the genocide, *Umuganda* was used to strengthen group cohesion within the "indigenous" ba-Hutu and marginalize the "non-indigenous" ba-Tutsi (Lawrence and Uwimbabazi, 2013). The patriotic focus of *Umuganda* became particularly salient in the early 1990's when "government propaganda gave no choice to Rwandans other than to attend Umuganda for political mobilization," (Lawrence and Uwimbabazi, 2013, p. 253).

Although little is known about the link between participation in *Umuganda* before the genocide and participation in violence during the genocide – a link which we hope to shed some light on in this paper – anecdotal evidence speaks to the importance of *Umuganda* as an instrument for local party and state officials to mobilize the peasant population. Since all Rwandans of working age, be it farmers or intellectuals, were required to participate in *Umuganda* (Guichaoua, 1991), the meetings were ideal for reaching the entire population. Although only a correlation, Straus (2006) shows that 88 percent of the perpetrators he interviewed regularly participated in *Umuganda* before the genocide.

Umuganda was also used during the genocide itself, with the new name *gukorn akazi*, or "do the work", which meant the killing of Tutsis (Verwimp, 2013). Other slogans related to *Umuganda* used before the genocide such as "clearing bushes and removing bad weeds" now had a completely altered connotation (Lawrence and Uwimbabazi, 2013). By equating participation in genocide violence with participation in *Umuganda*, the Hutu elite could signal that participation in genocide violence, just like participation in *Umuganda*, was a social obligation for all "true" Rwandans.

In 2008, the Tutsi-led government re-introduced *Umuganda* in Rwanda to promote development and reduce poverty in the aftermath of the genocide (Uwimbabazi, 2012). Participation is once more mandatory for all able-bodied individuals between 18 and 65 years of age, and typical tasks include cleaning streets, cutting grass and trimming bushes along roads, repairing public facilities or building houses for vulnerable individuals. The meetings now take place on the last Saturday of every month.

3 Data

We combine several datasets from various sources to construct our final dataset with a total of 1,433 Rwandan sectors. Sectors are the second smallest administrative level in Rwanda, and the level for which our outcome data on perpetrators is available. Table 1 reports the summary statistics for our variables.

Participation Rates Ideally, we would like to have a direct measure of participation rates. Since such data does not exist, we follow the literature and use prosecution rates for crimes committed during the genocide as a proxy (Friedman, 2013; Heldring, 2014; Rogall, 2014; Yanagizawa-Drott, 2014). Thus, we use a nation-wide sector-level dataset, provided by the government agency "National Service of Gacaca Jurisdiction", which collects the outcome of the almost 10,000 local Gacaca courts set up throughout the country to prosecute the genocidaires. Importantly, people were prosecuted in the sector where they committed their crimes and did not have to be physically present to be prosecuted. The courts identify two categories of perpetrators that we distinguish between in our analysis.

The first category, which we refer to as "organized perpetrators", includes people that were either leaders and organizers, or committed particularly brutal crimes. Most of these perpetrators either belonged to the army or the militia or were local leaders. Approximately 77,000 cases were handled in this category.⁸

The second category, which we refer to as "civilian perpetrators", includes accomplices rather than leaders and instigators. People accused in this category were not members of any of the organized groups mentioned for the first category and are thus considered to be civilians. Approximately 430,000 cases were recorded in this category. Our analysis includes 416,000 of these.⁹

⁸According to the precise definition taken from the National Service of Gacaca Jurisdiction this category concerns: (i) planners, organizers, instigators, supervisors of the genocide; (ii) leaders at the national, provincial or district level, within political parties, army, religious denominations or militia; (iii) well-known murderers who distinguished himself because of the zeal that characterized him in the killings or the excessive wickedness with which the killings were carried out; (iv) people who committed rape or acts of sexual torture.

⁹According to the precise definition taken from the National Service of Gacaca Jurisdiction this category concerns: (i) authors, co-authors, accomplices of deliberate homicides, or of serious attacks that caused someone's death; (ii) persons who – with the intention of killing – caused injuries or committed

As mentioned, the second category is our main outcome variable. In our analysis, we normalize the number of civilian prosecutions by sector Hutu population (Figure 1).¹⁰

The reliability of the prosecution data is an important issue for our analysis. In particular, if the prosecution process induced systematic errors our results may be biased. For instance, in sectors with a lot of violence, the killings might have been so widespread that no witnesses were left to accuse the perpetrators, thus resulting in low prosecution rates. Another concern is that *Umuganda* meetings did not lead to increased participation in violence, but rather only made the prosecution process easier because people knew whom to accuse. To address these concerns, we show that our results are robust to using an alternative measure of genocide violence from a different data source: the presence of mass graves. Our results are also similar when dropping those sectors with mass graves in the baseline regressions.

In addition, Friedman (2013) shows that the Gacaca data is positively correlated with several other measures of genocide violence from different sources.¹¹ More precisely, she states that "the Gacaca courts have been very thorough in investigating, and reports of those afraid to speak are rare, so this data is likely to be a good proxy for the number of participants in each area," Friedman (2013, pp. 19-20).

Another concern is that some of the people prosecuted in the Gacaca courts might not have committed their crimes during the genocide, but rather during the civil war preceding the genocide (October 1990 until August 1993). In particular, we cannot rule out that (a) some perpetrators may, in fact, have been accused of participation in massacres and other kinds of violence during the civil war (and not during the genocide), and (b) that individuals who had previously participated in violence during the civil war were more likely to have been recognized and trialed for genocide crimes than individuals who "only" participated in the genocide. In order to mitigate this concern, we exclude communes with violence against the Tutsi during the period October 1990 to

other serious violence, but without actually causing death; (iii) persons who committed criminal acts or became the accomplice of serious attacks, without the intention of causing death.

¹⁰Figure A.1 in the Appendix maps the number of civilian prosecutions without normalization.

¹¹These sources include a 1996 report from the Ministry of Higher Education, Scientific Research and Culture (Kapiteni, 1996); the PRIO/Uppsala data on violent conflicts (Gleditsch et al, 2002); and a database of conflict from Davenport and Stam (2009).

March 1994 (Viret, 2010). Importantly, violence against the Hutu population was not trialed in the Gacaca courts (Human Rights Watch, 2011; Longman, 2009).

Rainfall Data As our source of exogenous weather variation, we use rainfall data from the National Oceanic and Atmospheric Administration (NOAA) database of daily rainfall estimates, which stretches back to 1984.¹² The NOAA data combines actual weather station data with satellite information to obtain rainfall estimates at 0.1-degree (\sim 11 kilometers at the equator) latitude-longitude intervals. This data has two important advantages. First, since Rwanda is a small country, the high spatial resolution is crucial for obtaining reasonable variation in rainfall. Second, the high temporal resolution, i.e. daily estimates, allows us to confine the variation in rainfall to the exact days of Umuganda. Given Rwanda's hilly terrain, there is considerable local variation in rainfall. Moreover, these sectors criss-cross the various rainfall grids and each sector polygon is likely to overlap with more than one rainfall grid. The overall rainfall in each sector is thus obtained through a weighted average of the grids, where the weights are given by the relative areas covered by each grid. Naturally, there will always be some measurement error in the satellite data. However, as long as this error is classical it should only work against our findings. Moreover, in the following analysis we show that our results only hold using rainfall during the 3.5 years before the genocide (and no other 3.5 year period between 1984 and 2016). This rules out systematic errors in the rainfall data.

Sector Boundary, Road and City Data A sector boundary map is provided by the Center for Geographic Information Systems and Remote Sensing of the National University of Rwanda (CGIS-NUR) in Butare. Importantly, the map comes with information on both recent and old administrative groupings. Since Rwandan sectors have been reorganized under different higher administrative units several times after 1994, this information allows us to match sectors across the datasets (e.g. the 1991 census and the Gacaca records).

¹²Technically, the rainfall data is already available for 1983. However, for that year rainfall information is missing for several days and we therefore do not use it.

Africover provides spatial maps with major cities and roads derived from satellite imagery. The Africover project is part of the FAO's assistance to the Nile Basin countries in mapping out infrastructure. The maps are used to calculate several distance measures, such as the distance of the sector to the nearest main road, to the nearest city, to the country borders and to Kigali and Nyanza, the recent capital and the old Tutsi Kingdom capital, respectively. This data is also used to calculate sector areas.

Additional Data In addition to the data described above, we use census data on population size, ethnic composition, and radio ownership. These data are retrieved from Genodynamics and the IPUMS International census data base from 1991.¹³ Except for the population data, these variables are only available at a more aggregated (commune) level than the sector, which is our unit of analysis.¹⁴ We define ethnicity as the share of people that are Hutu or Tutsi, respectively. Importantly, the Tutsi minority is spread out across the whole of Rwanda with an average population share of about 10 percent. We calculate the Tutsi minority share used in the following analysis as the share of Tutsis normalized by the share of Hutus. As a measure of inter-ethnic connectedness we also calculate Inter-ethnic Marriage Polarization: $\sum_{i=1}^{F_c} \frac{N_{ic}}{N_c} \cdot h_{ic} \cdot t_{ic}$, where N_c is the total number of married people in all households F_c in commune c, N_{ic} the number of married people in household i and h_{ic} is the fraction of married household members in household i that are Hutu and t_{ic} the fraction that are Tutsi, respectively. The higher this measure is, the higher is the inter-ethnic connectedness.

Verpoorten (2012b) provides data on the location of mass graves based on satellite maps from the Yale Genocide Studies Program. Guichaoua (1991) provides information on the party affiliation of the commune leaders (called burgomasters) at the eve of the genocide.

The Terrain Ruggedness Index is obtained from Nunn and Puga (2012). Using elevation data at 30 arc-seconds (~ 0.9 kilometers at the equator) latitude-longitude grid cells, ruggedness is calculated in the following way. Let $e_{r,c}$ denote elevation at

¹³No census data was collected between 1991 and the genocide. However, mobility between sectors was highly limited because of governmental restrictions and land market controls (Prunier, 1995). This implies that the population data from 1991 is informative also of the situation in the wake of the genocide.

¹⁴The commune is one administrative level above the sector. There are 1433 sectors and 142 communes in our data. Thus, a commune includes on average 10 sectors.

the point located in row r and column c of a grid of elevation points. Then the Terrain Ruggedness Index at that point is given by $\sum_{i=r-1}^{i=r+1} \sum_{j=c-1}^{j=c+1} (e_{i,j} - e_{r,c})^2$. The elevation data source is GTOPO30 (US Geological Survey, 1996), a global elevation data set developed through a collaborative international effort led by staff at the US Geological Survey's Center for Earth Resources Observation and Science (EROS).

Yanagizawa-Drott (2014) provides data on the share of each sector that received the RTLM radio signal as well as the distance of each sector to the closest out of the two RTLM transmitters in the country and the mean and variance of sector elevation. The so-called "hate radio station" RTLM, set up in July 1993, broadcasted anti-Tutsi propaganda.

Matching of data and summary statistics The different datasets are matched by sector names within communes. A commune (142 in total) is the administrative unit above the sector. Unfortunately, the matching is not perfect: some sectors have different names in different data sources, and in some cases, two or more sectors within the same commune have identical names, which prevents successful matching. Nonetheless, in total, only about five percent of the sectors do not have a unique match across all datasets. Furthermore, these issues are likely idiosyncratic, which means that they will simply result in a lower precision in the estimates than in the case of perfect matching.

4 Empirical Strategy

To identify the effect of *Umuganda* meetings on participation in genocide violence, and because we lack data on the number of people participating in *Umuganda*, we use heavy rainfall as an instrument for low participation and estimate the reduced-form effect of heavy rainfall on civilian genocide participation. Our identification strategy thus rests on two assumptions. First, sectors with heavier rainfall on Saturdays experienced fewer or less intensive *Umuganda* meetings (first stage). Second, conditional on our control variables, rainfall on Saturdays does not have any direct effect on genocide violence other than through the *Umuganda* meetings (exclusion restriction).

First Stage Ideally, we would like to directly test the first-stage relationship using data on the number of people participating in *Umuganda* before the genocide. Unfortunately, such data does not exist. There is, however, abundant anecdotal evidence from today's Rwanda for expecting a strong first stage. Articles in the Rwandan media report low attendance at *Umuganda* meetings on days with heavy rain. One Rwandan newspaper writes that during a *Umuganda* meeting in Kigali's Gitega sector, participation was very limited because it started raining just as *Umuganda* started (Makuruki, 2015). According to our data, total rainfall in the area that day was 11 mm. Official government reports also document how the rains affect *Umuganda*. In Remera and Bwishyura sectors (in the southeast and the west of Rwanda), *Umuganda* was stopped because it started to rain (NDOW, 2016 and KDOW, 2016). Total rainfall on those days was 9 and 16 mm, respectively. Similarly heavy rainfalls affected most *Umuganda* meetings in Gicumbi district (in the north of Rwanda) with rainfall that day ranging from 9 to 15 mm (GDOW, 2015).¹⁵

Several other studies have documented and exploited negative relationships between rainfall and participation in open-air events. One of the first examples is Collins and Margo (2007) who use rainfall in April 1968 as an instrument for participation in the US riots after the death of Dr Martin Luther King. More recent examples include Madestam et al. (2013) and Madestam and Yanagizawa-Drott (2011). Similarly, several other studies use rainfall and other weather phenomena for an exogenous variation in voter turnout on election days (Eisinga et al., 2012; Gomez et al., 2012; Fraga and Hersh, 2011; Hansford and Gomez, 2010; Horiuchi and Saito, 2009).

However, in all these cases, rainfall has an effect both on the direct cost of attending the open-air event and the opportunity cost of attending. For example, Lind (2014) finds that voter turnout in Norway increases when it rains on the election day because bad weather reduces the opportunity cost of going to the polling station. Since *Umuganda* was mandatory, the opportunity cost mechanism is, however, unlikely to play any role. Instead, rainfall made the meetings and the work less productive, or led to cancella-

¹⁵As another example of how rainfall affects open-air events, a Rwandan newspaper reports that President Kagame's visit to Rwimiyaga sector was canceled due to heavy rains in the sector (Kigali Today, 2016). Rainfall that day was 15 mm.

tions. Still, the true functional form between rainfall and participation in mandatory community work is unknown.

To make progress, we assume – consistent with the anecdotal evidence – that the typical *Umuganda* tasks, consisting exclusively of outdoor work, became difficult or impossible to perform once a certain rainfall threshold had been reached.¹⁶ Following Harari and La Ferrara (2013), who define an extreme weather shock as two standard deviations from the long-term average, we choose this threshold to be 10 mm.¹⁷ Thus, we use the number of Saturdays from October 1990 to March 1994 with rainfall above 10 mm as our main explanatory variable.¹⁸ The starting date of our period of interest coincides with the outbreak of a civil war in Rwanda. We choose this starting date because it marks the beginning of a period of increased tensions between the two ethnic groups. Figure 2 shows that there is significant local variation in the number of rainy Saturdays for our period of interest.¹⁹ Furthermore, in Figure 3 we plot the number of rainy Saturdays after netting out commune fixed effects and average rainfall controls. This map further confirms that there is substantial local variation and little spatial clustering.

To better understand whether rainfall affected the extensive or the intensive margin of *Umuganda* meetings, we can vary these rainfall thresholds. More specifically, we also report the results when using thresholds between 5 and 15 mm (in increments of 1 mm). If we see effects already at low thresholds, it speaks for less enjoyable meetings or an effect at the intensive margin. If the effects are only seen at higher levels, cancellations are more likely to be driving the results, i.e. an effect at the extensive margin.

¹⁶The typical *Umuganda* tasks took place outside and, as mentioned above, included landscaping, road maintenance, construction and agriculture (Guichaoua, 1991).

¹⁷The long-term average daily rainfall in Rwanda from 1984 to 1994 was 2.6 mm with a standard deviation of 3.8 mm. We calculate this number taking the average across all sectors and all days from 1984 to 1994. Two standard deviations from the long-term average correspond to 10.24 mm. In Table A.1 in the Appendix, we show that our results are also robust to using average daily rainfall on Saturdays and all other weekdays.

¹⁸Madestam et al. (2013) use a threshold of 0.1 inches (2.5 mm) of rainfall, a light drizzle, to predict participation in the Tea Party Tax Day rally in the US. While a 2.5 mm threshold may be appropriate to capture participation in a voluntary rally in the US, we believe that our case, mandatory meetings, requires a higher threshold. Madestam et al. (2013) also use 0.35 inches (\approx 9 mm) as a robustness check for a higher threshold of rainfall. In Figure 5, we show that our results are also robust to using this threshold.

¹⁹In Figure A.2 in the Appendix we further show that there is significant local rainfall variation for all other weekdays.

Exclusion Restriction Our empirical strategy relies on the counterfactual assumption that, absent the *Umuganda* meetings, rainfall on Saturdays had no effect on genocide violence. This is unlikely the case without further precautions. Rainfall on Saturdays, like on all other weekdays, is likely to affect rain-fed production and is therefore correlated with income. Income, in turn, potentially affects genocide participation since participation was often driven by material incentives and because higher income enabled people to pay bribes to avoid participation in the genocide (Hatzfeld, 2005). Besides affecting agricultural outcomes, heavy rainfall might destroy infrastructure such as roads or housing, which also affects households' economic well-being and, therefore, their likelihood of participating in conflict.

To address this problem, and to isolate the Saturday rainfall effect, we control for average daily rainfall from January 1984 to September 1990 and our period of interest from October 1990 to March 1994. Furthermore, we control for sector Hutu population and rainfall on all other six weekdays. The absence of systematic, significant effects for days other than Saturdays serves as a first placebo test. In particular, this placebo test allows us to rule out that any other factors correlated with rainfall, such as geography, are driving the results because these factors should then naturally be correlated with rainfall on all other days as well. Nevertheless, we show, for instance, that our results are robust to controlling for sector ruggedness. To account for local characteristics, we also add 142 commune fixed effects.²⁰

At this point, we still need to argue that no other events potentially happening parallel with *Umuganda* on Saturdays could be driving our results. In particular, one might be concerned that people meeting and interacting in general might affect the participation in genocide violence. Although we cannot directly test for this, we will provide several indirect tests alleviating this concern.

²⁰Furthermore, in the Appendix we split the sample into sectors with high Saturday rainfall and low Saturday rainfall (split at the median) and report summary statistics for all of our exogenous variables for both sub-samples. Table A.2 confirms that there are no significant differences between the two samples.

Specifications We run the following reduced-form regression to estimate the effect of *Umuganda* meetings on participation in genocide violence

(1)
$$\frac{G_{ic}}{H_{ic}} = \alpha + \beta \# Saturdays (Rainfall > t mm)_{ic} + \mathbf{X}_{ic}\pi + \gamma_c + \varepsilon_{ic},$$

where G_{ic} is the number of Hutus prosecuted in each of the two categories of perpetrators (civilians or organized), i.e. our proxy for participation in genocide violence, and H_{ic} is the Hutu population in sector *i* in commune *c*. #*Saturdays*(*Rainfall* > *t mm*)_{*ic*} is our explanatory variable of interest: the number of Saturdays from October 1990 to March 1994 with rainfall above t mm. Our main specification uses 10 mm as a measure of heavy rainfall, but our results are robust to using other rainfall thresholds. X_{ic} is a vector of sector-specific controls, including sector Hutu population, average daily rainfall from January 1984 to September 1990, average daily rainfall from October 1990 to March 1994 and the number of all other weekdays with rainfall above t mm during our period of interest, October 1990 to March 1994. Finally, γ_c are commune fixed effects, and ε_{ic} is the error term. We allow the error terms to be correlated across sectors within the same commune by clustering the standard errors on the commune level. For the sake of robustness, we also allow error terms to be correlated across sectors within districts and within a 25, 50, and 75 km radius (Conley, 1999).²¹ Moreover, since the prosecution rates are heavily skewed to the right, we weight our observations by total sector population size, but our results do not rely on this weighting scheme. The coefficient of interest, β , captures the percentage point change in genocide participation following an additional Saturday with rainfall above t mm (we measure participation rates in percent).

5 Results

Main Effects The reduced-form relationship between the share of civilian perpetrators and the number of Saturdays with rainfall above 10 mm is strongly negative and

 $^{^{21}}$ The results are reported in Table A.3 in the Appendix. Districts were introduced after the genocide and we only report results here for the sake of robustness.

statistically significant at the 99 percent significance level (regression 1 in Table 2) and this relationship holds up when adding 142 commune fixed effects (regression 2) and the number of other weekdays with rainfall above 10 mm (regression 3). Regarding magnitude, the point estimate of -0.321 (standard error 0.109, regression 3 with all controls) suggests that a one standard-deviation increase in the number of rainy Saturdays reduces the civilian participation rate by 1.3 percentage points (note that the civilian participation rate is measured in percent).²² If we assume a one-to-one relationship between the number of rainy Saturdays and the number of canceled *Umuganda* meetings, a one standard-deviation increase in the number of canceled the average civilian participation rate by about 16 percent (interpreted at the mean share of civilian perpetrators, which is around 8 percent). Reassuringly, none of the other weekdays is systematically and significantly related to civilian violence (we cannot reject the null that these coefficients are all equal to zero, p-value 0.952).

Furthermore, to rule out that multicollinearity between rainfall on the different weekdays might be hiding otherwise significant effects, we rerun our main specification using daily rainfall above 10 mm for each weekday in a separate regression. Figure 4 confirms that only Saturday rainfall is significantly related to civilian participation in violence. The coefficients for rainfall on all other weekdays are much smaller in magnitude and statistically insignificant.²³

To understand whether rainfall led to cancellations, or rather lowered the *Umuganda* meeting attendance, we vary the threshold in increments of 1 mm: from 5 to 15 mm. Figure 5 reports the results. Heavy rainfall on Saturdays is negatively related to civilian participation for all thresholds above 5 mm and significant at least at the 90 percent confidence level for all thresholds between 10 and 14 mm. The fact that the strongest effects are observed for thresholds above 9 mm suggests that cancellations rather than lower attendance (or less enjoyable meetings) led to the decrease in violence in places

²²A standard deviation in the number of rainy Saturdays is 4.24.

²³In our main specifications, we control for long-term and general rainfall effects by including average daily rainfall before and during our period of interest. Another approach would be to use rainfall shocks, i.e. deviations of Saturday rainfall from long-term averages normalized by their long-term standard deviation. In Table A.4 in the Appendix, we show that our results are robust to using linear Saturday rainfall shocks as well as sector-specific rainfall thresholds (recall that in our main specification this Rwanda-wide threshold is 10 mm): The effects for Saturday rainfall are always highly statistically significant and similar in magnitude, while rainfall on the other weekdays do not seem to matter.

with a larger number of rainy Saturdays. Note that for higher thresholds the magnitude of the effects decreases again (in absolute value). This is not surprising since when using these high thresholds, days where *Umuganda* was canceled will be wrongly assigned to the control group and thus create a non-random measurement error.

It is important to note that since we use variation only at a high number of *Umu-ganda* meetings actually taking place (on average there are 18 Saturdays with rainfall above 10 mm, which corresponds to about 10 percent of all Saturdays in our period of interest), the above coefficient does not allow us to back out the counterfactual number of perpetrators if all meetings had been canceled. In particular, it is likely that the effects at the lower, out-of-sample end would be smaller if learning effects of *Umuganda* only arise after a while; for example, it might require a few meetings before the elites' propaganda convinces the population. Such potential non-linearities in the relationship between canceled *Umuganda* meetings and genocide participation would speak against using a simple linear extrapolation. To nevertheless get an idea of the magnitude of our results, we might consider the hypothetical situation of all sectors having 32 meetings canceled (the maximum in our sample, which amounts to about 8 months). Our main result suggests that the number of civilian perpetrators would then have fallen by around 50 percent.

To provide additional suggestive evidence that learning and practice seemed to have mattered, we show that *Umuganda* was especially successful in areas with many Saturday meetings in a row. In other words, in localities that did not have consecutive cancellations due to rain. To this end, we count the number of times a sector had two or three rainy Saturdays in a row and interact this number with the number of rainy Saturdays. Since there are almost no places with four rainy Saturdays in a row, we also count the number of 4-week and 5-week periods where a sector had at least three rainy Saturdays. The periods will then always start and end with a rainy Saturday. The results, presented in Table 3, show that the coefficients of all interaction terms are negative. However, for both two and three rainy Saturdays in a row, the effects are statistically insignificant (regressions 1 and 2). One explanation is that two rainy Saturdays may be too short of an interruption to disturb any learning effects of *Umuganda*, and unfor-

tunately there are very few sectors with three rainy Saturdays in a row, which leaves us with too little variation to be able to draw any conclusions. Reassuringly though, we find significant effects for the interaction with the number of 4-week periods with at least three rainy Saturdays. The point estimate of -0.202 (standard error 0.116) suggests that a one standard-deviation (0.4) increase in the number of rainy Saturdays in a row decreases the effectiveness of *Umuganda* by 25 percent (regression 3). Finally, allowing for two non-rainy Saturdays in the 5-week periods weakens the results (regression 4).

Reliability of the Gacaca Data and Robustness Checks Next, we perform a number of robustness checks and placebo tests, reported in Table 4. First, to rule out that some systematic error in the prosecution data is biasing our results, we use an alternative measure of genocide violence from a different data source: the location of mass graves (obtained through satellite images). Consistently, regressions 1 and 2 show that sectors with many rainy Saturdays are less likely to have a mass grave site altogether. The point estimate of -0.012 (standard error 0.004, regression 2) is significant at the 99 percent level. Moreover, the reduced-form point estimates are essentially identical to our baseline results and similarly significant at the 99 percent confidence level when dropping sectors with at least one mass grave (indicating high death rates, regression 3). This finding rules out that, for example, survival bias is driving our results, or that *Umuganda* did not induce higher participation, but rather made the prosecution process easier.

The results are also unaffected by adding a number of additional controls that potentially affect civilian participation in violence (regression 4). These include sector ruggedness, distance to the border, distance to cities, distance to Kigali and distance to Nyanza as well as Hutu population density. To illustrate why these can potentially affect our outcome: sector ruggedness is likely to be correlated with sector rainfall and to capture micro-climate effects; being close to the border potentially made it easier for the Tutsi or moderate Hutus to leave the country; the distance to cities, in particular the capital Kigali, is likely to be correlated with economic activity and public goods provision; and Nyanza was the old Tutsi Kingdom capital and sectors further away from it still exhibit lower Tutsi shares, on average. Lastly, Hutu population density is meant to capture social pressure as well as food pressure, both important reasons for the genocide (Verpoorten, 2012a).²⁴

The results are also robust to functional form. In regressions 5 and 6 we logaritmize our dependent variable.²⁵ Once more, the point estimates on Saturday rainfall is highly significant at the 99 percent level while none of the other weekdays matter. Moreover, the point estimate of -0.036 (standard error 0.012, regression 6) suggests that a one standard-deviation increase in the number of rainy Saturdays decreases civilian participation by about 15 percent (similar to the magnitude from our baseline results above).

In the spirit of a placebo check, the results for organized perpetrators are small and insignificant (regressions 7 to 8). This is not surprising: since organized perpetrators mainly consisted of members of the militia or the army, it is unclear that the sector where they committed their genocide crimes (and where they were subsequently prosecuted) is the same as the one where they lived – and were exposed to *Umuganda* – before the genocide. Thus, they may not have been exposed to the same number of *Umuganda* meetings as the inhabitants of that sector. If this is the case, when using organized perpetrator share as the outcome, our data is likely to suffer from measurement error.

To test for outliers, we also drop one commune at a time. The resulting estimates range from -0.280 to -0.364 and are significantly different from zero at the 99 percent confidence level in all cases (Figure A.3 in the Appendix). In addition, Figure 6 graphically shows that the negative relationship between the number of rainy Saturdays and civilian violence is not driven by any outliers (and neither is the null result for all other weekdays (Figure A.4 in the Appendix)).

Placebo Check As another placebo check, we rerun the main specification using Saturday rainfall during the 3.5 year pre-genocide calendar period (October 1, YEAR to March 31, YEAR+4), using data from all years in the period 1984-2016. To illustrate this, we begin with the period from October 1, 1984 until March 31, 1988 and end with the period from October 1, 2012 until March 31, 2016. Reassuringly, the coefficient

²⁴The food pressure argument assumes a Malthusian model: a fixed amount of agricultural land feeds a growing population (fertilizers were hardly used in Rwanda (Percival and Homer-Dixon, 2001)).

²⁵To account for observations with no violence we add a 1 to the dependent variable.

on Saturday rainfall from 1990 to 1994, the actual pre-genocide period, is an extreme outlier to the left in the distribution of the resulting 23 coefficients: none of the other point estimates are larger in absolute value (the result is shown in Figure 7).²⁶

Exclusion Restriction After demonstrating a strong and robust effect of high Saturday rainfall on civilian participation in genocide, we still have to argue that this effect results from people *participating in Umuganda*.

Most importantly, since major outdoor events, such as music festivals or soccer games, usually take place on weekends and are potentially affected by rainfall, one might be concerned that people meeting and interacting in general could affect participation in genocide violence. However, recalling our main result in Table 2, we find no significant effect for Sunday rainfall. Since people traditionally attend church on Sundays, this is the first piece of evidence speaking against social meetings in general driving our results. Furthermore, our results are robust to dropping the capital Kigali and other major cities in the sample; places where one might expect these major outdoor events to predominantly take place (regressions 1 and 2 in Table 5).

In a similar vein, heavy rainfall on public holidays, another occasion for people to meet, does not seem to matter: the point estimate on the number of public holidays with rainfall above 10 mm is statistically insignificant and small (regression 3 in Table 5; holiday variables are normalized by their standard deviation).²⁷ The same is true when adding religious and non-religious holidays separately to the regression (regression 4).²⁸

Throughout our period of interest from 1990 to 1994, violent acts against Tutsis and moderate Hutus were already taking place. If these pre-genocide perpetrators are included in the Gacaca data, and there is a relationship between rainfall before the

 $^{^{26}}$ In Figure A.5 in the Appendix we rerun the above placebo check using alternative thresholds to calculate our variables of interest. In particular, we use 5 to 15 mm thresholds in increments of 1 mm. We also report results for a linear specification with average weekly rainfall. Consistent with the results from Figure 5, the data passes the placebo test for all thresholds between 7 and 14 mm and for average rainfall.

²⁷Note that we exclude holidays that fall on a Saturday since these might still have been subject to *Umuganda*.

²⁸Religious holidays are, for instance, Easter and Christmas. Non-religious holidays in Rwanda are, for instance, Independence Day and Labor Day.

genocide and targeted violence during that period, for instance through transport costs, our estimates would be biased. To rule out this possibility, we drop communes where violence against the Tutsi took place before the genocide (Viret, 2010). Reassuringly, our results for civilian participation are robust (regression 5).

Another concern is that our estimates might be picking up effects from rainfall during the genocide rather than the 3.5 years pre-genocide period. To address this concern we control for the number of Saturdays (and all other weekdays) with rainfall above 10 mm during the period of the genocide (regression 6). Once more, our results hold up.

To provide further evidence that the effects we measure above result from the political elites abusing *Umuganda* meetings, we split the sample of sectors into areas with local pro-Hutu party leaders and opposition party leaders (the location of these sectors can be seen in Figure 8).²⁹ Interestingly, the negative relationship from above seems to be entirely driven by the pro-Hutu governed areas. The point estimate on Saturday rainfall is -0.388 (standard error 0.101, regression 8), which is slightly larger than our main effect and again highly significant at the 99 percent confidence level. The opposite seems to be true in opposition-governed areas: The point estimate on Saturday rainfall is large and positive, albeit given the small sample of only 161 sectors, it is insignificant (0.717, standard error 0.798, in regression 9 and 0.415, standard error 0.764, in regression 10 with all other weekday controls).

However, one concern is that the positive point estimates for opposition areas might be driven by larger Tutsi populations rather than by the identity of local leaders, since localities with opposition parties in power had, on average, larger Tutsi populations. To rule this out, we restrict the sample to sectors with large local Tutsi populations.³⁰ Reassuringly, the coefficient on Saturday rainfall turns negative and gets close to our baseline result for the full sample (regression 11).

As a final test, in Table A.5 in the Appendix, we pool the pro-Hutu and opposition samples and add an interaction term between heavy Saturday rainfall and a dummy tak-

²⁹The opposition parties include the Liberal Party (PL) and the Social Democratic Party (PSD). The pro-Hutu parties were the National Revolutionary Movement for Development (MRND) and the Republican Democratic Movement (MDR).

 $^{^{30}}$ We use sectors with Tutsi minority shares above 22 percent. This number is chosen to match the number of observations in regression 10.

ing on the value of one if the local leader is from an opposition party. For robustness sake, we do this separately for each threshold in the interval between 8 mm to 12 mm. While the coefficient on Saturday rainfall – the effect in pro-Hutu sectors – is negative and highly significant throughout and similar to our baseline result, the interaction term with the opposition leaders is positive, and, with only a few exceptions, statistically significant. On the other hand, the interaction term between Saturday rainfall and the Tutsi minority share are mostly insignificant and even negative throughout. This provides further support that the differential effects are due to political leadership as opposed to the size of the Tutsi minority.

Although only suggestive, taken at face value, the numbers imply that in these opposition sectors, the meetings were used to create bonds between the two ethnicities.

6 Channels

In this section, we attempt to better understand the channels and mechanisms behind the effect of *Umuganda* on civilian violence. Since we lack the detailed data to answer this question unambiguously, we take an agnostic approach and present suggestive empirical evidence.

6.1 Tracing the Effects over Time

In the preceding section we presented suggestive evidence that practice and repetition seemed to have mattered for *Umuganda's* effectiveness. To understand whether the observed effects result from long-term or short-term mobilization, we trace the effects over time.

First, we consider *Umuganda* before our period of interest,³¹ that is, the period of January 1984 to September 1990. Results, reported in Table 6, suggest that Saturday rainfall during the 1980s and beginning of the 1990s did not affect participation in violence. This is true both when we consider the entire period and when we split it into three equal-length time intervals. The coefficients are close to zero and insignificant

³¹Recall that our period of interest is October 1, 1990 until March 31, 1994.

throughout (regressions 1 and 2). Before offering an interpretation, we take a closer look at our period of interest.

Regression 3 shows that our main results are robust to controlling for Saturday rainfall during the pre-civil war period: 1984 to 1990 (split into 3 intervals). Next, we split our period of interest into three equal-length time intervals (14 months each). The results suggest that the effects are mainly driven by Saturday rainfall just before the genocide (regression 4). Splitting our pre-genocide/civil war period into 6 time intervals (each with a length of 7 months) confirms the previous result that the time right before before the genocide mattered the most (regression 5). A further split into 12 intervals shows that the last three coefficients on Saturday rainfall are highly significant, suggesting that the last 11 months before the genocide mattered the most. This result is similar when adding our additional controls (regression 6).³² Interestingly, the time around which we start observing significant results coincides with the launch of radio RTLM, which lead to a further radicalization of the country (Yanagizawa-Drott, 2014). In a final step, we estimate results separately for the periods before and after RTLM. As expected, we see the strongest effects for the time after RTLM (regressions 8 and 9).³³

Taken together, the results in this subsection suggest that the mechanism behind our main results is short-term mobilization of civilians during meetings rather than long-term effects of *Umuganda* on civilian participation working through e.g. improvements of the road system (recall that this was one of the major *Umuganda* tasks). In the next subsection, we provide a more detailed interpretation.³⁴

³⁴In the Appendix, we analyze the effects over time for pro-Hutu and opposition-governed areas separately. For pro-Hutu governed areas, the results mirror the general results from above, with the strongest

³²Note that with 42 months in total, 3, 6 and 12 are natural divides.

³³The fact that effects are stronger after the launch of RTLM could support an alternative explanation for our findings: If people only listened to the radio on Saturdays and if heavy rainfall reduces radio signal strength (radio fade), our results may be driven by RTLM rather than by the *Umuganda* meetings. However, there are several reasons why this is unlikely to explain our findings. Firstly, there are anecdotal accounts that in the period just before and during the genocide, people listened to RTLM all the time. An example is the following quote by Des Forges, interviewed regarding RTLM: "*People listened to the radio all the time, and people who did not have radios went to someone else's house to listen to the radio,*" Rwanda Initiative (2004). Yet, we find no effects of rainfall on our outcome variable for weekdays other than Saturdays. Secondly, we find opposite effects in opposition-governed sectors. Thirdly, radio fade is not only driven by rainfall in each sector but rather by rainfall along the way between the radio transmitter and the sector. In Table A.6 in the Appendix, we control for rainfall along the way between each sector and the transmitters (we count both the number of days that average rainfall exceeds a given threshold and the number of days that rainfall anywhere along the way exceeds that threshold). Importantly, our main results are robust to including these controls.

6.2 Heterogeneous Effects

To gain a deeper understanding of possible mechanisms, in this subsection we investigate whether there are heterogeneous effects depending on factors such as the party affiliation of the local leadership or the composition of the population in a sector. Since the mechanisms in pro-Hutu governed areas and opposition-governed areas are likely to differ, we analyze these two sub-samples separately. Building on the findings presented in the previous sub-section, we also separately examine the effects of Saturday rainfall before and after the launch of RTLM for the pro-Hutu governed areas. The small number of opposition-governed areas makes us unable to identify such effects in the opposition areas. The results for pro-Hutu governed areas are reported in Tables 7 and 8, while the results for opposition-governed areas are presented in Table 9. To ease comparison, all interaction variables are normalized by their standard deviations.

Pro-Hutu Sectors Starting with the pro-Hutu governed sectors, a natural first question is whether the political Hutu elites mostly used *Umuganda* meetings to spread propaganda and inform civilians about the views of the pro-Hutu government – something a radio reporter might have done just as well – or whether the *Umuganda* meetings worked on top of political propaganda by allowing local elites to bring civilians together, and thus functioned as a coordination device. If the *Umuganda* meetings mostly functioned as an information device the effect of canceled *Umuganda* meetings, i.e. Saturday rainfall, should be less negative (more muted) in sectors that were already informed through the radio.

We have two measures of access to information: a) the fraction of Hutus that owned a radio,³⁵ and b) the fraction of the sector that received the RTLM radio signal.³⁶ If

³⁶The RTLM coverage data and propagation control variables are taken from Yanagizawa-Drott

effects appearing just before the start of the genocide (Table A.7). For opposition-governed areas, the effects are positive for the time before RTLM and negative thereafter (Table A.8). However, sample sizes are, in this case, small and the results are at best suggestive. Nevertheless, for the sake of robustness, we can reproduce these results using alternative thresholds (Table A.9). Our interpretation of these suggestive results is that, with the launch of RTLM, the hate messages and anti-Tutsi propaganda intensified to a level that was difficult for the local opposition leaders to overcome. Unfortunately, due to the small sample size, we cannot split our period of interest into 12 or even 6 shorter intervals.

³⁵Two radio stations existed in Rwanda in the period before the genocide: Radio Rwanda and Radio RTLM, the former having national coverage. Both informed listeners about the pro-Hutu view of the government but Radio RTLM was propagating more aggressively against the Tutsis.

Umuganda mostly worked through information we should observe a positive interaction effect of Saturday rainfall with radio ownership or RTLM coverage, respectively. If *Umuganda* and the radio broadcasts were complementing each other, we should instead observe a negative interaction effect.

The point estimate on the interaction term with radio ownership is indeed positive (0.046, regression 1 in Table 7), but small and statistically insignificant (standard error 0.062). The point estimate on the interaction term with RTLM is negative, but also far from conventional significance levels (regression 2, Table 7). However, since RTLM started broadcasting in July 1993, only meetings after that date could have been affected by RTLM coverage. Splitting our period of interest accordingly, this is exactly what we observe. The interaction term between RTLM coverage and the number of rainy Saturdays between July 1993 and March 1994 is negative (-0.147, standard error 0.085) and significant at the 90 percent level (regression 2, Table 8). In terms of magnitude, the effects suggest that a one standard-deviation increase in RTLM coverage increased Umuganda's positive effect on violence by about 38 percent. Interestingly, we still find no effects for Hutu radio ownership (regression 1 in Table 8). One explanation for observing statistically significant interaction effects with RTLM coverage, but not with radio ownership, may be that the latter also picks up wealth effects. It could also be the case that the point estimate on radio ownership reflects the combined effect of having access to both radio stations, with Radio Rwanda being much less radical than RTLM, and that any effect of RTLM is therefore diluted. Nonetheless, these results suggest that Umuganda and the RTLM broadcasts were complements in the mobilization process. Note that the interaction effects for Hutu radio ownership are robust to adding all the other interaction terms (regressions 6 to 8 in Table 7 and Table 8).³⁷

One concern is that the radio signal might be weaker on days with heavy rainfall, meaning that our negative interaction effects might simply show that rainfall reduced the effects of RTLM. However, since villagers listened to RTLM all-week-around we

^{(2014).} Since Yanagizawa-Drott's sample is smaller than ours, we lose a number of observations.

³⁷Note further that we do include the interaction with RTLM coverage in the regressions with all interaction effects. This is because RTLM coverage is exogenous (see Yanagizawa-Drott (2014) for details) and thus unlikely to bias any results (or be biased itself). Besides, we would lose valuable observations.

should then also find a negative interaction for all other weekdays. Table A.10 in the Appendix shows that this is not the case.

To analyze more directly whether *Umuganda* was used by the local elites to bring people together, thus working as a coordination device, we next consider the interaction of Saturday rainfall with Hutu population density. Results, presented in regression 3 in Table 7, provide support for the coordination mechanism. The coefficient on the interaction term is positive and highly significant at the 99 percent confidence level. The point estimate of 0.093 (standard error 0.023) suggests that a one standard-deviation increase in the Hutu population density reduced the effects of *Umuganda* by about 24 percent. Thus, *Umuganda* appear to have been particularly effective in sparsely populated areas. Once more, the effect seems to be driven by rainfall after the launch of RTLM (regression 3 in Table 8) and it is robust to adding the other interaction terms (regressions 6 to 8 in Table 7 and Table 8).

The effectiveness of *Umuganda* might also depend on the size of the Tutsi minority. It is possible that large or well-integrated Tutsi minorities boycotted or hindered the effectiveness of *Umuganda* meetings. To examine this, we interact the Saturday rainfall with the number of Tutsis relative to the number of Hutus in each sector, and with the share of inter-ethnic marriages within each sector (a measure of inter-ethnic connectedness), respectively. The results, presented in regressions 4 to 8 in Table 7, speak against such a mechanism. The point estimates on the interaction terms between Saturday rainfall and the Tutsi population share/connectedness measure are insignificant and, if anything, negative.

This is perhaps not surprising since the Tutsi were the clear minority in Rwanda, never holding the majority in any sector, and therefore their presence presented no real threat to the activities of the Hutu elites. In fact, taken at face value, the negative point estimates suggest that the meetings were more successful in sectors with larger and better integrated Tutsi minorities. The perceived Tutsi threat might have been more salient in these sectors and the enemy easier to point out. Being threatened by "the Tutsi enemy" should have been especially salient after the launch of RTLM, and consistently we find large and mostly significant, negative interaction effects of the Tutsi minority

size/connectedness with Saturday rainfall during the time period after the launch of RTLM (regressions 4 to 8 in Table 8).

Opposition Sectors As a final step, we present suggestive evidence for areas with opposition party leadership. Because of the small number of opposition-governed sectors the results in this subsection should be interpreted with caution.

First, in these sectors, it seems that the local elites used the *Umuganda* meetings to compensate for the anti-Tutsi propaganda spread on the radio. The interaction effect of Saturday rainfall with radio ownership among the Hutus and RTLM coverage are both negative, and the latter is significant at the 95 percent confidence level (Table 9, regressions 1 and 2). The point estimate on RTLM coverage of -2.264 (standard error 0.999) suggests that the positive effect of *Umuganda* is about 1.5 times lower in areas with a RTLM coverage level of one standard deviation as compared to areas with no coverage at all (regression 2). Once more, the results in Table A.10 in the Appendix suggest that this effect is not due to heavy rainfall reducing radio signal strength. Further note that, in this sub-sample, the interaction effects for Hutu radio ownership are robust to adding all other interaction variables and even turn statistically significant (regressions 6 to 8).

Interactions of Saturday rainfall with population density do not show the same pattern as for the pro-Hutu governed areas. The interaction term in regression 3 is small and statistically insignificant (-0.177, standard error 0.529). This is also the case once we add all other interaction variables (regressions 6 to 8).

Finally, the local opposition leaders seem to have been more effective in sectors with larger Tutsi minorities. The interaction of Saturday rainfall with inter-ethnic marriage is positive and, except for one specification, statistically significant (regressions 5, 7, and 8). This is consistent with the opposition leaders having to overcome a potential pro-Hutu bias among the Hutu population (which potentially was less pronounced in integrated sectors). Also note that the sheer size of the Tutsi minority does not deliver a clear picture. Point estimates are always positive (regressions 4, 6, and 8), but turn statistically insignificant once inter-ethnic marriage is controlled for (regression 8).

Summary Summing up, our results in this subsection indicate that in pro-Hutu governed areas *Umuganda* meetings and the RTLM broadcasts were complements in the mobilization process and meetings were most effective in sparsely populated areas, thus functioning as a coordination device. In opposition-governed areas, our results suggest that local elites used the *Umuganda* meetings to compensate for the anti-Tutsi resentments spread on the radio or present in the Hutu population.

7 Discussion and Conclusion

This paper explores how an authoritarian state can mobilize civilians to participate in mass violence through a strong top-down policy. The policy, *Umuganda*, required Rwandan citizens to attend politically infused work meetings every Saturday in the years before the 1994 genocide. We map out the causal effect of *Umuganda* on genocide participation by using exogenous weather fluctuations on each Saturday, which would affect participation in the work meetings, and compare it to all other weekdays.

Our results show that the local Hutu elites used these mandatory community meetings to mobilize the civilian population for genocide. More precisely, we find that a one standard-deviation increase in the number of rainy Saturdays (*Umuganda* days) decreased the share of civilian perpetrators in the Rwandan Genocide by around 16 percent. This effect is entirely driven by areas ruled by the pro-Hutu parties, and having more meetings in the last 11 months before the genocide seems to have been particularly important for mobilization.

Our findings are important for several reasons. First, a large number of civilians participated in the killings during the Rwandan Genocide. While it is a common understanding that this genocide was centrally planned and organized, little is known about the link between the planning and the wide acceptance of the genocide among the civilian population. Our paper suggests that weekly held community meetings played a major role in this preparation and mobilization process.

Second, the Rwandan Genocide is part of a wider phenomenon of civil conflict and social unrest. Understanding its mobilization dynamics is important for interpreting and handling other events where civilian actors matter, such as the 2007 post-election vio-

lence in Kenya where "communities turned on each other with crude weapons as they were encouraged, and even paid, by power-hungry politicians." (BBC, 2010). Moreover, Brown (1999) suggests that the vast majority of civil conflicts is elite-triggered.

Third, people getting together during community meetings is commonly said to foster a sense of belonging and create social capital, generally viewed as positive for development and community building (Knack and Keefer, 1997; Grootaert and van Bastelaer, 2002; Guiso, Sapienza and Zingalez, 2008). As emphasized by Putnam (2000), social capital can bridge the divides in a society. However, we show that there is a "dark side" to these community meetings. More specifically, our results indicate that when placed in the wrong hands, the effects can become disastrous.

A more optimistic view of this institution might explain why the current Rwandan government reinstalled *Umuganda* in 2008: When placed in the right hands, it could (partly) work against propaganda and overcome hatred. Indeed, official statements about *Umuganda* emphasize values such as "solidarity" and "reconciliation", and the practice is said to foster a sense of community. These mandatory work days are now held monthly, on the last Saturday of every month. A similar practice is also present in Burundi and is being discussed in the Democratic Republic of Congo and Kenya. Our analysis clearly shows that these meetings are powerful instruments and warrant caution, not least in countries with histories of ethnic tensions.

References

- BBC. Kenya election violence: ICC names suspects, 2010.
 http://www.bbc.co.uk/news/world-africa-11996652, accessed on October 21, 2016.
- [2] Besley, T. and T. Persson. The Logic of Political Violence, *Quarterly Journal of Economics*, 126(2011), pp. 1411-1445.
- [3] Blattman, C. and E. Miguel. Civil War, *Journal of Economic Literature*, 48(2010), pp. 3-57.
- [4] Brown, M. E. (ed.). *The International Dimensions of Internal Conflict*, MIT Press, Cambridge, Massachusetts, 1996.
- [5] Brückner M. and A. Ciccone. Rain and the Democratic Window of Opportunity, *Econometrica*, 79(2010), pp. 923-947.
- [6] Collins W. J. and R. A. Margo. The Economic Aftermath of the 1960s Riots in American Cities: Evidence from Property Values. *The Journal of Economic History*, 67(2007), pp. 849-883.
- [7] Conley, T. G. GMM Estimation with cross sectional Dependence, *Journal of Econometrics*, 92(1999), pp. 1-45.
- [8] Cook, S. E. (Ed.). *Genocide in Cambodia and Rwanda: new perspectives*, No. 1, Transaction Publishers, 2004.
- [9] Dallaire, R. Shake hands with the devil, Random House Canada, Toronto, 2003.
- [10] Daily Nation. Emuhaya MP Seeks to Introduce Mandatory Community Cleaning Day, 2016, http://allafrica.com/stories/201603030209.html, accessed on October 21, 2016.
- [11] Davenport, C. and A. Stam. Rwandan Political Violence in Space and Time, mimeo, 2009.

- [12] Dell, M. Trafficking Networks and the Mexican Drug War, American Economic Review, 105(2015), pp. 1738-1779.
- [13] Des Forges, A. Leave None to Tell the Story: Genocide in Rwanda, Human Rights Watch and the International Federation of Human Rights Leagues, New York, NY, USA, 1999. www.hrw.org/legacy/reports/1999/rwanda/.
- [14] Dube, O. and J. F. Vargas. Commodity Price Shocks and Civil Conflict: Evidence from Colombia, *Review of Economic Studies*, 80(2013), pp. 1384-1421.
- [15] Eisinga, R., M. Grotenhuis and B. Pelzer. Weather Conditions and Political Party Vote Share in Dutch National Parliament Election, 1971-2010, *International Journal of Biometeorology*, 56(2012), pp. 1161-1165.
- [16] Fraga, B. and E. Hersh. Voting Costs and Voter Turnout in Competitive Elections, *Quarterly Journal of Political Science*, 5(2011), pp. 339-356.
- [17] Friedman, W. Local Economic Conditions and Participation in the Rwandan Genocide, mimeo, 2013.
- [18] Gicumbi District Official Website (GDOW). Umuganda Usoza Ukwezi Kwa Gashyantare, 2015, http://www.gicumbi.gov.rw/index.php?id=38&tx_ ttnews%5Btt_news%5D=104&cHash=16ab35c781b1fdc2b6c0cc7db450d207, accessed on October 21, 2016.
- [19] Gleditsch, N. P., Wallensteen, P., Eriksson, M., Sollenberg, M. and H. Strand. Armed Conflict 1946-2001: A New Dataset, *Journal of Peace Research*, 29(2002), pp. 615-637.
- [20] Gomez D. P., Aronow, P. M. and M. C. McGrath. Field Experiments and the Study of Voter Turnout, *Journal of Elections, Public Opinion & Parties*, 23(2012), pp. 27-48.
- [21] **Gourevitch, P.** We wish to inform you that tomorrow we will be killed with our *families*, Farrar, Straus & Giroux, New York, 1998.

- [22] Grootaert, C. and T. van Bastelaer. Understanding and Measuring Social Capital: A Multi-Disciplinary Tool for Practitioners, Washington, World Bank, 2002.
- [23] Guichaoua, A. Les Travaux Communautaires en Afrique Centrale, *Revue Tiers Monde*, t.XXXII, n. 127, July-September, 1991, pp. 551-573.
- [24] Guiso, L., Sapienza P. and L. Zingales. Alfred Marshall Lecture: Social Capital as Good Culture, *Journal of the European Economic Association*, 6(2008), pp. 295-320.
- [25] Hansford, T. G. and B. T. Gomez. Estimating the Electoral Effects of Voter Turnout, American Political Science Review, 104(2010), pp. 268-288.
- [26] Harari, M. and E. La Ferrara. Conflict, Climate and Cells: A Disaggregated Analysis, C.E.P.R. Discussion Papers 9277, 2013.
- [27] Hatzfeld, J. Machete season: The Killers in Rwanda speak, Picador, New York, 2005.
- [28] Heldring, L. State Capacity and Violence: Evidence form the Rwandan Genocide, CSAE Working Paper, WPS/2014-08, 2014.
- [29] Horiuchi, Y. and J. Saito. Rain, Election, and Money: The Impact of Voter Turnout on Distributive Policy Outcomes, Yale, mimeo, 2009.
- [30] Human Rights Watch. Justice Compromised The Legacy of Rwanda's Community-Based Gacaca Courts, Report, May 2011.
- [31] **IRDP.** Sustaining Peace in Rwanda: Voice of the People, Institut de Recherche et de Dialogue pour la Paix, 2003.
- [32] Iyer, L. and P. Topalova. Poverty and Crime: Evidence from Rainfall and Trade Shocks in India, mimeo, 2014.
- [33] Jones, B. and B. Olken. Hit or Miss? The Effect of Assassinations on Institutions and War, *American Economic Journal: Macroeconomics*, 1(2009), pp. 55-87.
- [34] Kapiteni, A. La Première Estimation du Homre des Victims du Genocide du Rwanda de 1994 Commune par Commune en Fev 1996, Report of the Ministry of Higher Education, Scientific Research, and Culture, 1996.
- Offical [35] Karongi District Website (KDOW). Umuganda Rusange kwa gashyantare, 2016, http://www.karongi. wa usoza ukwezi gov.rw/index.php?id=38&tx_ttnews[tt_news]=198&cHash= 855ee589e4e7a07c9ec87a9c1c8e2ccb, accessed on October 21, 2016.
- [36] Kigali Today. Abaturage bababajwe nâisubikwa ryâuruzinduko rwa Perezida, 2016, http://www.kigalitoday.com/amakuru/mu-rwanda/ Abaturage-bababajwe-n-isubikwa-ry-uruzinduko-rwa-Perezida, accessed on October 21, 2016.
- [37] Knack, S. and P. Keefer. Does Social Capital Have an Economic Payoff? A Cross-Country Investigation, *Quarterly Journal of Economics*, 112(1997), pp. 1251-1288.
- [38] Lawrence, R. and P. Uwimbabazi. Indigenous Practice, Power and Social Control: The Paradox of the Practice of Umuganda in Rwanda in Race, Power and Indigenous Knowledge Systems, *Interdisciplinary Journal for the Study of the Arts* and Humanities in Southern Africa, 20(2013), pp. 248-272.
- [39] Lind, J. T. Rainy Day Politics: An Instrumental Variables Approach to the Effect of Parties on Political Outcomes, University of Oslo, mimeo, 2014.
- [40] Longman, T. An Assessment of Rwanda's Gacaca Courts, *Peace Review*, 21(2009), pp. 304-312.
- [41] Madestam A., Shoag, D., Veuger S. and D. Yanagizawa-Drott. Do Political Protests Matter? Evidence from the Tea Party Movement, *Quarterly Journal of Economics*, 128(2013), pp. 1633-1685.
- [42] Madestam A. and D. Yanagizawa-Drott. Shaping the Nation: The Effect of Fourth of July on Political Preferences and Behavior in the United States, mimeo, 2011.

- [43] Makuruki. MU MAFOTO: Uko Umuganda wakozwe mu bice bimwe by'Umujyi wa Kigali, henshi haranzwe ubwitabire buke cyane, 2016, http://www.makuruki.rw/Politiki/article/MU-MAFOTO-Uko-Umugandawakozwe-mu-bice-bimwe-by-Umujyi-wa-Kigali-henshi-haranzweubwitabire-buke-cyane, accessed on October 21, 2016.
- [44] Mamdani, M. When Victims Become Killers: Colonialism, Nativism, and the Genocide in Rwanda, Fountain, Kampala, 2001.
- [45] Mansuri, G. and V. Rao. Localizing development: does participation work?, World Bank Publications, 2012.
- [46] Melvern, L. A People Betrayed: The Role of the West in Rwanda's Genocide, Zed-Books, London, 2000.
- [47] Melvern, L. Conspiracy to Murder: The Rwandan Genocide, Verso, New York, 2004.
- [48] Miguel, E., Satyanath, S. and E. Sergenti. Economic Shocks and Civil Conflict: An Instrumental Variables Approach, *Journal of Political Economy*, 112(2004), pp. 725-753.
- [49] Mitra, A. and D. Ray. Implications of an Economic Theory of Conflict: Hindu-Muslim Violence in India, *Journal of Political Economy*, 122(2014), pp. 719-765.
- [50] **Mukarubuga, C.** The Experience of Social Forums Against Poverty: The Case of Rwanda, Agency for Co-Operation and Research in Development, 2006.
- [51] Ngoma District Offical Website (NDOW). Abayobozi bâakarere bakoreye Umuganda i Remera, 2016, http://www.ngoma.gov.rw/index.php?id=38& tx_ttnews[tt_news]=334&cHash=523c816de256ba76a47cdc002ed6d9b4, accessed on October 21, 2016.
- [52] Nunn, N. and D. Puga. Ruggedness: The blessing of bad geography in Africa, *Review of Economics and Statistics*, 94(2012), pp. 20-36.

- [53] Nunn, N. and N. Qian. U.S. Food Aid on Civil Conflict, American Economic Review, 104(2014), pp. 1630-1666.
- [54] Percival, V. and T. Homer-Dixon. Environmental Scarcity and Violent Conflict: The Case of Rwanda, *The Journal of Environment and Development*, 5(1996), pp. 270-291.
- [55] Putnam, R. D. Bowling Alone, Free Press, New York, 2000.
- [56] Pottier, J. Land Reform for Peace: Rwanda's 2005 Land Law in Context, *Journal of Agrarian Change*, 6(2006), pp. 509-537.
- [57] Prunier, G. The Rwanda Crisis: History of a Genocide, Hurst and Company, London, 1995.
- [58] Rogall, T. Mobilizing the Masses for Genocide, University of British Columbia, mimeo, 2015.
- [59] Rwanda Initiative. Hate Media in Rwanda, 2004, http://www. rwandainitiative.ca/symposium/transcript/panel1/questions.html, accessed October 27, 2016.
- [60] Sarsons, H. Rainfall and Conflict: A Cautionary Tale, *Journal of Development Economics*, 115(2015), pp. 62-72.
- [61] Satyanath, S., Voigtlaender, N. and H.-J. Voth. Bowling for Fascism: Social Capital and the Rise of the Nazi Party, *Journal of Political Economy*, forthcoming.
- [62] Straus, S. The Order of Genocide: Race, Power, And War in Rwanda, Cambridge University Press, 1 edition, 2006.
- [63] US Geological Survey. GTOPO30, Sioux Falls, SD: United States Geological Survey Center for Earth Resources Observation and Science (EROS), 1996.
- [64] Uvin, P. Aiding Violence: the Development Enterprise in Rwanda, Kumarian Press, 1998.

- [65] Uwimbabazi, P. An Analysis of Umuganda: The Policy and Practice of Community Work in Rwanda, College of Humanities, KwaZulu-Natal, South Africa, September 2012.
- [66] Verpoorten, M. Leave None to Claim the Land: A Malthusian Catastrophe in Rwanda?, *Journal of Peace Research*, 49(2012a), pp. 547-563.
- [67] Verpoorten, M. Detecting Hidden Violence: The Spatial Distribution of Excess Mortality in Rwanda. *Political Geography*. 31(2012b), pp. 44-56.
- [68] Verwimp, P. Development Ideology, the Peasantry, and Genocide: Rwanda Represented in Habyarimana's Speeches, *Journal of Genocide Research*, 2(2000), pp. 325-361.
- [69] Verwimp, P. Machetes and Firearms: the Organization of Massacres in Rwanda, *Journal of Peace Research*, 43(2006), pp. 5-22.
- [70] Verwimp, P. Peasants in Power: The Political Economy of Development and Genocide in Rwanda, Springer, Heidelberg, 2013.
- [71] Viret, E. Rwanda A Chronology (1867-1994), Online Encyclopedia of Mass Violence, published on 1 March 2010, accessed 27 October 2014, http://www. massviolence.org/Rwanda-A-Chronology, ISSN 1961-9898.
- [72] Yanagizawa-Drott, D. Propaganda and Conflict: Evidence from the Rwandan Genocide, *Quarterly Journal of Economics*, 129(2014), pp. 1947-1994.

Figures



Figure 1: Civilian Participation, normalized by Hutu Population



Figure 2: Local Variation in Saturday Rainfall



Figure 3: Local Variation in Saturday Rainfall – Residuals

Notes: The map plots the residuals of the number of Saturdays with rainfall above 10 mm after netting out commune fixed effects and our other average rainfall controls.



Figure 4: Effects of Rainfall by Weekday

Notes: The figure shows the coefficients on the number of days with rainfall above 10 mm (together with 95 percent confidence bounds) for each weekday entering our main specification separately (regression 2 in Table 2). Each point represents a separate regression.



Figure 5: Different Thresholds

Notes: The figure shows the coefficients on the number of rainy Saturdays (together with 95 percent confidence bounds) when varying the rainfall threshold in our main specification (regression 3 in Table 2). Each point represents a separate regression.



Figure 6: Saturday-Rainfall-Violence Relationship

Notes: Observations are grouped into 75 equal-sized bins. We use all controls in regression 3 in Table 2 to construct residuals.





Notes: The figure shows the distribution of coefficients on the number of rainy Saturdays when using Saturday rainfall during the 3.5 years of the pre-genocide calendar period (October 1, YEAR to March 31, YEAR+4) from the years 1984 to 2016 to construct our variable of interest in regression 3 in Table 2.



Figure 8: Location of Pro-Hutu and Opposition Sectors

Tables

Table 1:	Summary	Statistics
----------	---------	------------

A. Violence & Population # Civilian Perpetrators 290.25 286.43 1433 # Militiamen 51.76 70.51 1433 # Civilian Perpetrators per Hutu (p.H.) 7.66 7.94 1433 # Militiamen per Hutu (p.H.) 1.41 2.10 1433 Pre-Genocide Violence against Tutsi, dummy 0.15 0.36 1433
Civilian Perpetrators 290.25 286.43 1433 # Militiamen 51.76 70.51 1433 # Civilian Perpetrators per Hutu (p.H.) 7.66 7.94 1433 # Militiamen per Hutu (p.H.) 1.41 2.10 1433 Pre-Genocide Violence against Tutsi, dummy 0.15 0.36 1433
Militiamen 51.76 70.51 1433 # Civilian Perpetrators per Hutu (p.H.) 7.66 7.94 1433 # Militiamen per Hutu (p.H.) 1.41 2.10 1433 Pre-Genocide Violence against Tutsi, dummy 0.15 0.36 1433
Civilian Perpetrators per Hutu (p.H.) 7.66 7.94 1433 # Militiamen per Hutu (p.H.) 1.41 2.10 1433 Pre-Genocide Violence against Tutsi, dummy 0.15 0.36 1433
Militiamen per Hutu (p.H.) 1.41 2.10 1433 Pre-Genocide Violence against Tutsi, dummy 0.15 0.36 1433
Pre-Genocide Violence against Tutsi, dummy 0.15 0.36 1433
Mass Grave found in Sector, dummy 0.05 0.21 1432
Hutu Population in Sector, '000 4.26 2.17 1433
Population in Sector, '000 4.88 2.48 1433
Hutu Population Density, per square km 425.07 605.37 1433
Population Density, per square km 498.60 850.61 1433
B. Rainfall
Sun(Rainfall>10 mm) 15.14 5.19 1433
Mon(Rainfall>10 mm) 15.13 4.22 1433
#Tue(Rainfall>10 mm) 18.10 3.52 1433
Wed(Rainfall>10 mm) 20.51 4.76 1433
Thu(Rainfall>10 mm) 21.53 3.97 1433
#Fri(Rainfall>10 mm) 17.02 4.75 1433
Sat(Rainfall>10 mm) 18.25 4.24 1433
Average Daily Rainfall, 1980s, mm 2.75 0.48 1433
Average Daily Rainfall, 1990s, mm 2.44 0.55 1433
Sun(Rainfall>10 mm), Genocide Period 0.32 0.48 1433
Mon(Rainfall>10 mm), Genocide Period 1.30 0.46 1433
Tue(Rainfall>10 mm), Genocide Period 1.42 0.72 1433
Wed(Rainfall>10 mm), Genocide Period 0.55 0.69 1433
Thu(Rainfall>10 mm), Genocide Period 0.00 0.07 1433
Fri(Rainfall>10 mm), Genocide Period 1.17 0.39 1433
Sat(Rainfall>10 mm), Genocide Period 0.05 0.22 1433
Pub. Holidays(Rainfall>10 mm) 0.85 0.20 1433
Non-Rel. Holidays(Rainfall>10 mm) 1.56 0.21 1433
Rel. Holidays(Rainfall>10 mm) 1.00 0.11 1433
C. Other Variables
Fraction of Hutu with Radio 0.33 0.09 1433
Tutsi Minority Share 0.10 0.13 1433
Inter-ethnic Marriage Polarization 0.01 0.01 1433
Ruggedness Index 3.45 1.38 1433
Distance to Kigali 62.65 30.00 1433
Distance to Main City 22.78 14.69 1433
Distance to Town 16.58 8.83 1433
Distance to Nyanza 64.36 30.74 1433
Distance to the Main Road 6.71 5.77 1433
Distance to the Border 22.60 13.93 1433
RTLM Coverage 0.19 0.22 1057
Distance to Transmitter 5.18 2.85 1057
Mean Altitude, km 1.71 0.23 1057
Variance in Altitude, m 9070.29 10370.73 1057

Notes: The # prosecuted militiamen is crime category 1: prosecutions against organizers, leaders, army and militia; # prosecuted civilians is crime category 2: prosecutions against civilians. The per Hutu (p.H.) variables are expressed in percent. Pre-Genocide Violence against Tutsi, is a dummy taking on the value of 1 if the sector experienced violence against Tutsi in the pre-genocide period. The two average daily rainfall variables are measured in millimeters. The distance variables are measured in kilometers. Population is the population number in the sector and Population Density is population per square kilometers, from the 1991 census. Radio ownership and ethnicity data are taken from the 1991 census, available only at the commune level. There are 142 communes in the sample. The Tutsi Minority Share is defined as the fraction of Tutsi normalized by the fraction of Hutu. The formulas for Inter-ethnic Marriage Polarization and the Ruggedness Index can be found in the Data Section in the paper. RTLM Coverage is the share of the sector that received the RTLM radio signal. Distance to Transmitter is the distance of the sector to the closest RTLM transmitter (there were two).

Dependent Variable:	# Civiliar	Perpetrators	, p.H.
	(1)	(2)	(3)
# Sat(Rainfall>10 mm)	-0.354 (0.115)***	-0.341 (0.105)***	-0.321 (0.109)***
# Sun(Rainfall>10 mm)			0.025 (0.092)
# Mon(Rainfall>10 mm)			(0.071) (0.097)
# Tue(Rainfall>10 mm)			(0.029) (0.075)
# Wed(Rainfall>10 mm)			(0.073) 0.018 (0.103)
# Thu(Rainfall>10 mm)			(0.103) -0.051 (0.112)
# Fri(Rainfall>10 mm)			(0.112) -0.043 (0.093)
Standard Controls	yes	yes	yes
Commune Effects	no	yes	yes
R ²	0.18	0.53	0.53
Ν	1433	1433	1433

Table 2: Main Effects

Notes: **# Sat(Rainfall>10 mm)** is the number of Saturdays with rainfall above 10 mm during the period October 1990 to March 1994 (and similarly for all other weekdays). **# Civilian Perpetrators per Hutu (p.H)** are measured in percent. **Standard Controls** include Hutu population, average daily rainfall for January 1984 to September 1990 and average daily rainfall for October 1990 to March 1994. All regressions are run using weighted least squares (WLS) estimation with Hutu population size as weights. There are **142 communes** in the sample. **Standard errors** are clustered at the commune level. *significant at 10 percent, **significant at 5 percent, ***significant at 1 percent.

Dependent Variable:	#	Civilian Perpe	etrators, p.H.	
	(1)	(2)	(3)	(4)
# Sat(Rainfall>10 mm)	-0.307 (0.139)**	-0.261 (0.118)**	-0.278 $(0.111)^{**}$	-0.330 (0.110)***
x # Seq. with 2 Rainy Saturdays in a Row	-0.003 (0.031)			
x # Seq. with 3 Rainy Saturdays in a Row	× /	-0.072 (0.090)		
x # 4-week Seq. with at least 3 Rainy Saturdays		. ,	$(0.116)^{*}$	
x # 5-week Seq. with at least 3 Rainy Saturdays			× /	$-0.088 \\ (0.240)$
Standard Controls	yes	yes	yes	yes
Other Weekday Controls	yes	yes	yes	yes
Commune Effects	yes	yes	yes	yes
\mathbb{R}^2	0.53	0.53	0.53	0.53
Ν	1433	1433	1433	1433

Table 3: Rainy Saturdays in a Row

Notes: **# Sat(Rainfall>10 mm)** is the number of Saturdays with rainfall above 10 mm during the period October 1990 to March 1994. **# Civilian Perpetrators per Hutu (p.H)** is measured in percent. As an example, **# 4-week Seq. with at least 3 Rainy Saturdays** is the number of 4 week sequences with at least 3 rainy Saturdays, i.e. rainfall above 10 mm (sequences start and end with a rainy Saturday). **Standard Controls** include Hutu population, average daily rainfall for January 1984 to September 1990 and average daily rainfall for October 1990 to March 1994. **Other Weekday Controls** include the number of Sun/Mon/Tue/Wed/Thu/Fri with rainfall above 10 mm during the period October 1990 to March 1994. All regressions are run using weighted least squares (WLS) estimation with Hutu population size as weights. There are **142 communes** in the sample. **Standard errors** are clustered at the commune level. *significant at 10 percent, **significant at 5 percent, ***significant at 1 percent.

Dependent variable:	Mass Grave	in Sector	# Civilian Perp	etrators, p.H.	Log [# Ci Perpetrator	vilian s, p.H.]	# Orgar Perpetrato	iized rs, p.H.
	Alterna Dep. V	tive /ar.	Without Mass Graves	Additional Controls	Transfor Dep. V	med /ar.	Othe Dep.	sr Var.
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)
# Sat(Rainfall>10 mm)	-0.014 $(0.004)^{***}$	-0.012 (0.004)***	-0.318 (0.111)***	-0.318 (0.113)***	-0.040 (0.013)***	-0.036 (0.012)***	-0.047 (0.030)	-0.040 (0.028)
# Sun(Rainfall>10 mm)		0.002	0.038	-0.001		0.004		-0.041
# Mon(Rainfall>10 mm)		(0.003) -0.002	(0.094) 0.050	(990.0) 0.111		(0.011)		(0.030) 0.086
# Tue(Rainfall>10 mm)		(0.004) 0.007	(0.095) 0.068	(0.097) 0.024		(0.013) -0.007		$(0.029)^{***}$ -0.036
		$(0.004)^{*}$	(0.079)	(0.076)		(0.012)		(0.028)
# Wed(Rainfall>10 mm)		0.006	0.014	0.00		0.002		0.013
		(0.004)	(0.100)	(0.098)		(0.013)		(0.027)
# Thu(Rainfall>10 mm)		-0.003	-0.049	-0.033		-0.030		-0.065
# Fri(Rainfall>10 mm)		(0.004) -0.008	-0.008	(0000-		(0.000) -0.009		0.006
		$(0.003)^{**}$	(0.091)	(0.093)		(0.011)		(0.025)
Standard Controls	yes	yes	yes	yes	yes	yes	yes	yes
Additional Controls	no	no	no	yes	ou	ou	no	ou
Commune Effects	yes	yes	yes	yes	yes	yes	yes	yes
\mathbb{R}^2	0.15	0.16	0.53	0.53	0.71	0.71	0.37	0.38
Ν	1432	1432	1367	1433	1433	1433	1433	1433

Table 4: Reliability of Gacaca Data and Robustness Checks

rates). In **regression 4** we add additional controls. In **regressions 5 and 6** we logaritmize the dependent variable. In **regressions 7 and 8** we use the number of organized perpetrators prosecuted in the sector as another dependent variable. **Standard Controls** include Hutu population, average daily rainfall for January 1984 to September 1990 and average daily rainfall for January 1984. Nyanza, the border, the closest main road and the closest main city and town. All regressions are run using weighted least squares (WLS) estimation with Hutu population size as weights. There are **142 communes** in the sample. **Standard errors** are clustered at the commune level. *significant at 10 percent, ***significant weekdays). # Civilian Perpetrators per Hutu (p.H) and # Organized Perpetrators per Hutu (p.H) are measured in percent. In regressions 1 and 2 we use a dummy indicating whether a mass grave was found in the sector as an alternative dependent variable. In regression 3 we drop sectors with mass graves (indicating high death at 1 percent.

Dependent variable:					# Civiliar	Perpetrators,	p.H.				
	Without Kigali	Without Major Cities	Public Ho	lidays	Excl. Pre- Violence	Genocide Rainfall	Local Pro Leade	-Hutu rs	Local Opp Leade	osition rrs	Large Tutsi Minority
	(1)	(2)	(3)	(4)	(2)	(9)	(7)	(8)	(6)	(10)	(11)
# Sat(Rainfall> 10 mm)	-0.341 (0.110)***	-0.347 (0.113)***	-0.319 (0.106)***	-0.308 (0.105)***	-0.416 (0.123)***	-0.284 $(0.110)^{**}$	-0.407 (0.096)***	-0.388 (0.101)***	0.717 (0.900)	0.415 (0.798)	-0.599 (0.764)
# Sun(Rainfall> 10 mm)	0.026	0.046	0.028	0.031	0.035	0.010		0.015		0.389	-0.336
# Mon(Rainfall> 10 mm)	0.065 0.065 0.065	0.049 0.049 0.099)	0.073	0.064 0.064 (0.097)	0.113	0.020 (0.096)		0.028		0.041 (0.366)	-0.586 (0.857)
# Tue(Rainfall> 10 mm)	0.021	0.035	0.034	0.037	-0.011	0.018		0.029		-0.008	0.237
# Wed(Rainfall> 10 mm)	-0.003	0.007	0.017	0.017	-0.018	0.037		-0.023		0.709	0.886
# Thu(Rainfall> 10 mm)	(0.108) -0.038	(0.115) -0.040	(0.105) -0.050	(0.102) - 0.049	(0.107) 0.082	(0.105) 0.017		(0.103) -0.116		$(0.400)^*$ 0.845	$(0.508)^{*}$ -0.445
	(0.115)	(0.119)	(0.112)	(0.113)	(0.126)	(0.119)		(0.107)		(0.615)	(0.458)
# FU(Kalman> 10 mm)	(0.093)	(0.098)	(0.106)	(0.119)	(0.108)	(0.095)		(0.089)		(0.419)	(0.604)
<pre># Pub. Holidays(Rainfall>10 mm)</pre>			-0.082 (0.433)								
# Non-Rel. Holidays(Rainfall>10 mm)			~	-0.214 (0.341)							
# Rel. Holidays(Rainfall>10 mm)				-0.576 (0.409)							
Standard Controls	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Genocide Controls	ou	ou	no	no	ou	yes	ou	no	ou	no	no
Commune Effects	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
K ² N	0.53 1422	0.53 1358	0.53 1433	0.53 1433	0.50 1213	0.53 1433	0.56 1266	0.56 1266	0.33 161	0.35 161	0.52 168
Notes: # Sat(Rainfall>10 mm) is the nurr Hutu (p,H) is measured in percent. In reg number of public holidays (separated into 1 the genocide. In regression 6 , we control f sectors in communes ruled by pro-Hutu par large Tutsi minority shares. Standard Con Controls include the number of Saturdays estimation with Hutu population size as w ***significant at 1 percent.	pher of Saturday ression 1 we dr religious and nc or the number o rties. In regressi ntrolla H with rainfall ab with rainfall ab	s with rainfall ab op sectors in the on-religious holid of Saturdays with tions 9 and 10, the out population, a utu population, a ove 10 mm durin	ove 10 mm duu capital Kigali i ays in regressi rainfall above s cample is restr verage daily ra g the period of es in the samp	ing the period and in regression on 4) with rair 00 mm during irfleted to sector infell for Janu the genocide le. Standard	I October 1990 ion 2 we drop tfall above 10 the genocide (the genocide (ary 1984 to Se (and similarly errors are clu	1 to March 199 all sectors in mm. In regre and similarly up s ruled by opp ptember 1990 for all other w istered at the o	14 (and similar and close to th saind close to the saind 5, we dry all other weekto all other weekto osition parties, and average d. veektays). All veektays). All	ly for all other e cities. In reg p sectors were flays). In regres lays, In regression aily rainfall for regressions are tegressions are	weekdays). # vestions 3 an violence aga sions 7 and 4 11 the sample October 1999 run using we at 10 percent	Civilian Per d 4 , we also d inst Tutsi tool <i>s</i> , the sample <i>s</i> is restricted to to March 19 to March 19 to March 19 sighted least <i>s</i> significans	petrators per control for the k place before is restricted to o sectors with 94. Genocide quares (WLS)

Restriction	
Exclusion	
Table 5:	

Dependent variable:				# Civilia	in Perpetrators,	p.H.			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
# Sat(Rainfall>10 mm) - Pre-Poli	tical Umuganda	Period							
Period 84-90: whole period	-0.016								
Period 84-90: 1. 27 Months	(0.057)	-0.238	-0.211	-0.227	-0.236	-0.258	-0.289	-0.253	-0.282
Period 84-90: 2. 27 Months		(0.166) 0.076	0.164)	(0.164) 0.111	(0.165) 0.124	0.112	(0.176) 0.110	0.118	(0.171) 0.117
Period 84-90: 3. 27 Months		(0.120) 0.024 (0.162)	(0.121) 0.068 (0.158)	(0.123) 0.069 (0.157)	(0.120) 0.074 (0.158)	(0.122) 0.110 (0.159)	(0.118) 0.094 (0.171)	(0.119) 0.099 (0.154)	(0.116) 0.082 (0.168)
# Sat(Rainfall>10 mm) - Political	Umuganda Perio	od							
Period 90-94: whole period			$(0.113)^{***}$						
Period 90-94: 1. 14 Months			()	-0.170					
Period 90-94: 2. 14 Months				(0.213) -0.059 (0.172)					
Period 90-94: 3. 14 Months				-0.592 (0.161)***					
Period 90-94: 1. 7 Months				· /	-0.141				
Period 90-94: 2. 7 Months					-0.094 (0.466)				
Period 90-94: 3. 7 Months					0.078				
Period 90-94: 4. 7 Months					-0.272 (0.250)				
Period 90-94: 5. 7 Months					-0.264				
Period 90-94: 6. 7 Months					(0.234) -0.840 $(0.233)^{***}$				
Period 90-94: 1. 3.5 Months					(0.255)	-0.299	-0.365		
Period 90-94: 2. 3.5 Months						(0.340) -0.147 (0.287)	-0.169		
Period 90-94: 3. 3.5 Months						(0.287) -0.177 (0.847)	0.111		
Period 90-94: 4. 3.5 Months						-0.220	-0.299		
Period 90-94: 5. 3.5 Months						(0.432) -0.443 (0.417)	-0.508		
Period 90-94: 6. 3.5 Months						0.559	0.522		
Period 90-94: 7. 3.5 Months						(0.427) -0.475 (0.658)	(0.427) -0.460 (0.662)		
Period 90-94: 8. 3.5 Months						-0.171	-0.181		
Period 90-94: 9. 3.5 Months						-0.130 (0.254)	(0.291) -0.046 (0.235)		
Period 90-94: 10. 3.5 Months						(0.234) -1.996 $(0.594)^{***}$	(0.233) -1.847 $(0.595)^{***}$		
Period 90-94: 11. 3.5 Months						(0.394) -0.939 $(0.278)^{***}$	(0.355) -0.843 $(0.285)^{***}$		
Period 90-94: 12. 3.5 Months						(0.276) -0.581 $(0.318)^*$	(0.203) -0.599 $(0.320)^*$		
Period 90-94: before RTLM						(0.510)	(0.520)	-0.099	-0.106
Period 90-94: after RTLM								(0.144) -0.889 $(0.219)^{***}$	-0.853 (0.216)***
Standard Controls Other Weekday Controls Additional Controls Commune Effects P ²	yes yes no yes 0.53	yes yes no yes 0.53	yes yes no yes 0.53	yes yes no yes 0.53	yes yes no yes 0.53	yes yes no yes 0.54	yes yes yes 0.54	yes yes no yes 0.54	yes yes yes yes
N	1433	1433	1433	1433	1433	1433	1433	1433	1433

Table 6: Mechanism: Effects over Time

Notes: All of the regressors in this table capture the number of Saturdays with rainfall above 10 mm for different time periods. **Regressors 1** is the number of rainy Saturdays for the period January 1984 to September 1990. **Regressors 2 to 4** are the number of rainy Saturdays for the period January 1984 to September 1990 (split into three time intervals of equal length). **All the other regressors** are from the time period October 1990 to March 1994 (split into various different time intervals of equal length, i.e. 3, 6, 12, and before/after RTLM). **# Civilian Perpetrators per Hutu (p.H)** is measured in percent. **Standard Controls** include Hutu population, average daily rainfall for January 1984 to September 1990 and average daily rainfall for October 1990 to March 1994. **Other Weekday Controls** include the number of Sun/Mon/Tue/Wed/Thu/Fri with rainfall above 10 mm during the period October 1990 to March 1994. **Other Weekday Controls** include the number of Sun/Mon/Tue/Wed/Thu/Fri with rainfall above 10 kgail, Nyanza, the border, the closest main road and the closest main city and town. All regressions are run using weighted least squares (WLS) estimation with Hutu population size as weights. There are **142 communes** in the sample. **Standard errors** are clustered at the commune level. *significant at 1 percent.

(2) -0.352 -		JVIIIAN Perpe	trators, p.m.			
-0.352 -	(3)	(4)	(5)	(9)	(2)	(8)
$(0.168)^{**}$ (0.463 $0.109)^{***}$	-0.288 (0.123)**	-0.275 (0.115)**	-0.452 (0.291)	-0.512 (0.281)*	-0.544 (0.303)*
				0.031 (0.074)	0.060 (0.074)	0.062 (0.077)
-0.035 (0.043)						
Ù	0.093 $0.023)^{***}$			0.086 $(0.030)^{***}$	0.108 $(0.028)^{***}$	0.116 (0.029)***
		-0.175 (0.186)		-0.226	х. У	0.180
		(00110)	-0.142 (0.096)		-0.234 (0.067)***	-0.341 (0.212)
yes	yes	yes	yes	yes	yes	yes
yes	yes	yes	yes	yes	yes	yes
yes	yes	yes	yes	yes	yes	yes
yes	no	ou	ou	no	no	ou
yes	yes	yes	yes	yes	yes	yes
0.54	0.56	0.56	0.56	0.57	0.57	0.57
921 1	266	1266	1266	1266	1266	1266
aturdays with rai aturdays with rai Minority Share i he Data Section i aily rainfall for O and uring the per the border, the clc the border, the clc variance of sector variances in the	fail above ship is the f s the fraction in the paper. rectober 1990 iod October sest main re- sest main re- sestmain re- settion. At	10 mm durin raction of Hu n of Tutsi div Standard C to March 19 1990 to Mai ad and the cl and and the cl madard error	g the period O tu owning a ra ided by the fra ottrols includd 094. Other WV och 1994. Add osest main city are run using s are clustered	tober 1990 to the RTLM of the thur of thutu. The the population of the population of the population of the population of the communication of the communic	March 1994. overage is the Inter-tehnic I tion, average d ols include the ols are sector LM Controls squares (WLS ne level. *sign	# Civilian fraction of <i>Marriage</i> is aily rainfall number of ruggedness, are distance o estimation fifcant at 10
yes 0.54 921 921 Minorit he Data aily rainf and durin the borde the borde the borde the borde the borde the borde	with rain with rain o Owner of Section in Section in all for O at the clc of sector of sector	yes 0.56 1266 with rainfall above o Ownership is the f Section in the paper. "all for October 1990 g the period October 1990 ctobes thain to of sector elevation. A nes in the sample. St	yes yes 0.56 0.56 1266 1266 with rainfall above 10 mm durin, o Ownership is the fraction of Hus div Section in the paper. Standard C Section in the paper. Standard C all for October 1990 to March 19 g the period October 1990 to March 19 g the period October 1990 to March 19 et he closest main road and the closest main of sector elevation. All regressions of sector elevation. All regressions	yes yes yes yes 0.56 0.56 0.56 1266 1266 0.56 with rainfall above 10 mm during the period O o Ownership is the fraction of Hutu owning a ra y Share is the fraction of Tutsi divided by the fra Section in the paper. Standard Controls include all for October 1990 to March 1994. Add at the previous of an of the closest main city of sector elevation. All regressions are run using nes in the sample. Standard errors are clustered	yes yes yes yes yes yes of 0.56 0.56 0.57 1266 0.56 0.56 0.57 1266 1266 1266 1266 1266 0.57 1266 1266 1266 1266 1266 1266 1266 126	yes yes yes yes yes yes yes yes of 0.56 0.56 0.57 0.57 1266 1266 0.56 0.57 0.57 1266 1266 1266 1266 1266 1266 1266 126

Table 7: Mechanism: Interaction Effects – Pro-Hutu Sectors

Ą
E
R
ffe
e/al
ore
Эef
s (1
OC
ect
L S
utt
Ŧ
50
Τ
cts
fe
臣
on
<u>icti</u>
era
Int
ü
uisr
nan
ect
Σ
ö
le
ab

Dependent variable:			#	Civilian Perp	etrators, p.H.			
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)
# Sat(Rainfall>10 mm), before RTLM	-0.626	-0.290	-0.296	-0.179	-0.116	-0.537	-0.547	-0.609
v Dodio Ownershin	$(0.352)^{*}$	(0.206)	$(0.128)^{**}$	(0.136)	(0.126)	(0.357)	(0.347)	$(0.364)^{*}$
	(0.086)					0.087)	(0.092)	(0.094)
x RTLM Coverage	~	0.041				~	~	~
x Hutu Population Density		(010.0)	0.072			0.072	0.097	0.105
. T. toi Minamiter Channel			(0.057)	0.001		(0.056)	$(0.055)^{*}$	$(0.057)^{*}$
				(0.181)		(0.172)		(0.334)
x Inter-ethnic Marriage					-0.144 (0.098)		-0.191 (0.088)**	-0.327 (0.228)
# Sat(Rainfall>10 mm), after RTLM	-0.584	-0.628	-0.881 (0.209)***	-0.498 (0.200)**	-0.625	-0.351 (0.530)	-0.504	-0.457
x Radio Ownership	-0.044			(00 - 0)		-0.061	-0.040	-0.038
x RTLM Coverage	(0.121)	-0.147				(0.1.0)	(0.148)	(0.145)
		$(0.085)^{*}$						
x Hutu Population Density			0.152 (0.083)*			0.174 (0.083)**	0.182 (0.085)**	0.172 (0.084)**
x Tutsi Minority Share				-0.490		-0.656 -0.656		-0.359
x Inter-ethnic Marriage				(017:0)	-0.156	(007.0)	-0.350	-0.170
					(c/.1.0)		$(0.125)^{***}$	(0.282)
Standard Controls	yes	yes	yes	yes	yes	yes	yes	yes
Other Weekday Controls	yes	yes	yes	yes	yes	yes	yes	yes
Additional Controls	yes	yes	yes	yes	yes	yes	yes	yes
RTLM Controls	ou	yes	ou	ou	ou	ou	ou	ou
Commune Effects	yes	yes	yes	yes	yes	yes	yes	yes
\mathbb{R}^2	0.57	0.54	0.57	0.57	0.57	0.57	0.57	0.57
Ν	1266	921	1266	1266	1266	1266	1266	1266
Notes: # Sat(Rainfall>10 mm), before (after) RTL March 1994). # Civilian Perpetrators per Hutu (p.) of the sector area that receives the RTLM signal. Th	M is the number I) is measured in Isi Minority Sh	r of Saturdays n percent. Rad are is the frac	with rainfall ab io Ownership tion of Tutsi di	ove 10 mm du is the fraction vided by the f	ring the period of Hutu ownin raction of Hutu	l October 1990 g a radio. RTL 1. Inter-ethnic	to June 1993 (M coverage is	July 1993 to the fraction
inter-ethnic connectedness defined in the Data Section 1990 and average daily rainfall for October 1990 to	n in the paper. March 1994. O	Standard Cor ther Weekday	utrols include H	Hutu population ude the numb	on, average dail er of Sun/Mon	ly rainfall for J n/Tue/Wed/Thu	lanuary 1984 to /Fri with rainf	o September all above 10
mm during the period October 1990 to March 1994.	Additional Co	ontrols are sec	tor ruggedness	, Hutu popula	tion density, di	istance to Kiga	uli, Nyanza, the	border, the
closest main road and the closest main city and town.	RTLM Contro	ls are distance	to the closest F	TLM transmi	tter as well as r	mean and varia	nce of sector e	evation. All
regressions are run using weighted least squares (WI	S) estimation w	ith Hutu popu	lation size as w	eights. There	are 142 comm	unes in the sar	nple. Standar	d errors are
clustered at the commune level. *significant at 10 pe.	cent, **significa	ant at 5 percen	t, ***significan	t at 1 percent.				

				•				
Dependent variable:				# Civilian Perp	etrators, p.H.			
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)
# Sat(Rainfall>10 mm)	3.349 (2.204)	1.524 (1.171)	0.238 (0.882)	-1.240 (0.905)	-1.015 (1.033)	2.117 (1.686)	5.085 (2.035)**	4.205 $(1.828)^{**}$
x Radio Ownership	-0.929 (0.607)					-0.966 (0.508)*	-2.117 (0.753)**	-2.007 $(0.723)^{**}$
x RTLM Coverage		-2.264 (0.999)**						
x Population Density		~	-0.177 (0.529)			-0.217 (0.563)	-0.130 (0.532)	-0.158 (0.537)
x Tutsi Minority Share				0.990 $(0.562)^{*}$		1.035 $(0.564)^{*}$		0.552 (0.549)
x Inter-ethnic Marriage					$0.814 \\ (0.674)$		$1.670 \\ (0.770)^{**}$	$(0.740)^{*}$
Standard Controls	yes	yes	yes	yes	yes	yes	yes	yes
Other Weekday Controls	yes	yes	yes	yes	yes	yes	yes	yes
Additional Controls	yes	yes	yes	yes	yes	yes	yes	yes
RTLM Controls	ou	yes	ou	no	ou	ou	no	ou
Commune Effects	yes	yes	yes	yes	yes	yes	yes	yes
\mathbb{R}^2	0.40	0.38	0.39	0.40	0.40	0.41	0.42	0.42
Z	161	132	161	161	161	161	161	161
Notes: # Sat(Rainfall>10 mm) is a Perpetrators per Hutu (p.H) is me the sector area that receives the RTL a measure of inter-ethnic connected for January 1984 to September 1990 Sun/Mon/Tue/Wed/Thu/Fri with rain Hutu population density, distance to to the closest RTLM transmitter as w with Hutu population size as weights percent, ** significant at 5 percent, **	the number of easured in perc M signal. Tuts ness defined in ness defined in and average nfall above 10 Kigali, Nyanza vell as mean and s. There are 14	Saturdays with ent. Radio Oye ia Minority Sh the Data Section daily rainfall f mm during the mm during the border, th i variance of se 2 communes i t 1 percent.	n rainfall abov vnership is th vnership is the fraction on in the pape or October 19 e period Octol e closest main e closest main e the sample.	<i>ie</i> 10 mm duri ie fraction of F tion of Tutsi di er. Standard di ber 1990 to March 1 ber 1990 to M rroad and the c . All regression	ng the period lutu owning a vided by the fi Controls inclu 994. Other V arch 1994. Ad losest main cit losest main cit ns are clustere ors are clustere	October 1990 radio. RTLM action of Hutu de Hutu popul Veekday Cont ditional Cont y and town. R weighted leas d at the comm	to March 1994 coverage is th coverage is th ation, average dation, average dation rols are sector TLM Controls t squares (WLS une level. *sign	# Civilian e fraction of Marriage is laily rainfall e number of ruggedness, are distance in estimation ificant at 10

Table 9: Mechanism: Interaction Effects - Opposition Sectors

Appendix – For Online Publication

A.1 Additional Figures



Figure A.1: Gacaca Data: Civilian Participation



Figure A.2: Local Rainfall Variation for all Other Weekdays

(a) Sunday Rainfall

(b) Monday Rainfall



(c) Tuesday Rainfall

(d) Wednesday Rainfall



(e) Thursday Rainfall

(f) Friday Rainfall



Figure A.3: Outliers

Notes: This figure shows the distribution of point estimates on the number of rainy Saturdays when dropping one commune at a time in our baseline specification (regression 3 in Table 2 in the paper).



Figure A.4: Relationship between Civilian Violence and Rainfall for all Other Weekdays

Notes: Observations are grouped into 75 equal-sized bins. We use all controls in regression 3 in Table 2 in the paper to construct residuals.



Figure A.5: Placebo Check – Different Rainfall Thresholds

Notes: The figure shows the distribution of coefficients on the number of rainy Saturdays when using Saturday rainfall during the 3.5 years of the pre-genocide calendar period (October 1, YEAR to March 31, YEAR+4) from the years 1984 to 2016 to construct our variable of interest. We vary the thresholds from 5 mm to 15 mm in increments of 1 mm. The last figure (bottom right) uses average Saturday rainfall.

A.2 Additional Tables

Dependent Variable:			#	Civilian Perpe	trators, p.H.			
	(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)
Average Rainfall Sat	-3.646 (1.379)***	-3.553 $(1.438)^{**}$						
Average Rainfall Sun		-0.084 (1.661)	-0.161 (1.618)					
Average Rainfall Mon		0.188 (1.426)	~	1.181 (1.686)				
Average Rainfall Tue		(1.112)			1.162 (1.368)			
Average Rainfall Wed		0.992			~	1.483 (1.249)		
Average Rainfall Thu		(1.061)					-0.749 (1188)	
Average Rainfall Fri		(0.873)						0.071 (0.775)
Standard Controls	yes	yes	yes	yes	yes	yes	yes	yes
Commune Effects	yes	yes	yes	yes	yes	yes	yes	yes
\mathbb{R}^2	0.53	0.53	0.52	0.53	0.53	0.53	0.53	0.52
Ν	1433	1433	1433	1433	1433	1433	1433	1433
Notes: Average Rainfall Sat is the Perpetrators per Hutu (p.H) is average daily rainfall for October There are 142 communes in the spercent.	he average daily Sat s measured in perce r 1990 to March 19 sample. Standard	urday rainfall du ant. Standard C 994. All regressi errors are cluste	ring the period fi ontrols include ons are run usin red at the comm	rom October 1990 Hutu population. g weighted least une level. *signi) to March 1994 (, average daily ra squares (WLS) e ficant at 10 perce	and similarly fo infall for Janua stimation with F nt, **significant	r all other weekda ry 1984 to Septer Jutu population s at 5 percent, ***	ys). # Civilian nber 1990 and ize as weights. significant at 1

Table A.1: Main Effects – Linear Specification

	Low Rainfall	High Rainfall	
	Mean (Std.dev.)	Mean (Std.dev.)	p-Value of Difference
Population in Sector, '000	5.035	4.729	0.248
	(2.394)	(2.561)	
Hutu Population in Sector, '000	4.435	4.092	0.275
	(1.938)	(2.376)	
Tutsi Population in Sector, '000	0.600	0.638	0.154
	(0.698)	(0.414)	
Population Density, per square km	610.455	386.897	0.595
	(1177.130)	(195.841)	
Hutu Population Density, per square km	522.167	328.106	0.545
	(830.348)	(159.941)	
Tutsi Population Density, per square km	88.287	58.791	0.935
	(355.393)	(56.559)	
Inter-ethnic Marriage Polarization	0.010	0.013	0.243
	(0.012)	(0.010)	
Sector Centroid Latitude, km	9781.806	9768.154	0.384
	(39.477)	(45.812)	
Sector Centroid Longitude, km	789.015	832.588	0.319
	(40.174)	(37.962)	
Distance to Kigali	67.074	58.241	0.451
C	(36.333)	(21.036)	
Distance to Main City	19.347	26.205	0.939
,	(10.652)	(17.165)	
Distance to Town	17.453	15.717	0.305
	(8.717)	(8.872)	
Distance to Nyanza	68.704	60.022	0.077
	(20.841)	(37.656)	
Distance to the Main Road	6.620	6.804	0.466
	(6.286)	(5.208)	
Distance to the Border	21.490	23.717	0.249
	(13,759)	(14.022)	0.2.0
Ruggedness Index	3.617	3.281	0.936
	(1.505)	(1.211)	0.700
RTLM Coverage	0.270	0.101	0.361
in Lin coverage	(0.264)	(0.130)	0.001
Distance to Transmitter	4,761	5.607	0.928
	(3.316)	(2.213)	
Mean Altitude, km	1.751	1.672	0.835
	(0.266)	(0.183)	0.055
Variance in Altitude, m	11506.127	6629.833	0.213
variance in a fittude, in	11500.127	5627.655	0.215

Table A.2: Summary Statistics by Rainfall on Saturdays

Notes: The full sample is split into two subsamples of equal size at the median value of average Saturday rainfall. The p-value of the test of equality of means is based on standard errors clustered at the commune level (and after netting out commune fixed effects). The distance variables are measured in kilometers. Hutu (Tutsi) population is the Hutu (Tutsi) population number in the sector and population density variables are measured as population per square kilometers, from the 1991 census. Latitude and Longitude (in km) are used to construct Conley standard errors in Table A.3. The formulas for Inter-ethnic Marriage Polarization and the Ruggedness Index can be found in the Data Section in the paper. RTLM Coverage is the share of the sector that received the RTLM radio signal. Distance to Transmitter is the distance of the sector to the closest RTLM transmitter (there were two). There are 142 communes in the sample.

Dependent Variable:	#	Civilian Perpe	etrators, p.H.	
	25 km	50 km	75 km	District
	(1)	(2)	(3)	(4)
# Sat(Rainfall>10 mm)	-0.316 [0.128]**	-0.316 [0.132]**	-0.316 [0.130]**	-0.321 [0.108]***
# Sun(Rainfall>10 mm)	0.034 [0.094]	0.034 [0.087]	0.034 [0.080]	0.025 [0.095]
# Mon(Rainfall>10 mm)	0.116 [0.094]	0.116 [0.104]	0.116 [0.105]	0.071 [0.102]
# Tue(Rainfall>10 mm)	0.047 [0.100]	0.047	0.047	0.029
# Wed(Rainfall>10 mm)	0.095 [0.106]	0.095	0.095	0.018
# Thu(Rainfall>10 mm)	-0.019 [0.120]	-0.019 [0.131]	-0.019 [0.144]	-0.051 [0.121]
# Fri(Rainfall>10 mm)	-0.059 [0.110]	-0.059 [0.105]	-0.059 [0.088]	-0.043 [0.097]
Standard Controls	yes	yes	yes	yes
Commune Effects	yes	yes	yes	yes
R ²	0.49	0.49	0.49	0.53
Ν	1433	1433	1433	1433

 Table A.3: Additional Standard Errors

Notes: **# Sat(Rainfall>10 mm)** is the number of Saturdays with rainfall above 10 mm during the period October 1990 to March 1994 (and similarly for all other weekdays). **# Civilian Perpetrators per Hutu (p.H)** is measured in percent. **Standard Controls** include Hutu population, average daily rainfall for January 1984 to September 1990 and average daily rainfall for October 1990 to March 1994. In **regressions 1 to 3 standard errors** correcting for spatial correlation within a radius of 25km, 50km and 75km are in square brackets, Conley (1999). The radius used in each regression is given in the column header. In **regression 4** we cluster at the district level. There are **142 communes** and **30 districts** in the sample. Regression 4 is run using weighted least squares (WLS) estimation with Hutu population size as weights. *significant at 10 percent, **significant at 5 percent, ***significant at 1 percent.

Dependent variable:				#	Civilian Perpe	trators, p.H.				
							Thresholds or	1 Average		
					y = x	mn	x = 10	um	x = 11	mm
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)	(6)	(10)
Saturday Rainfall Shock	-23.967 $(8.025)^{***}$	-24.169 (9.502)**	-22.667 $(7.700)^{***}$	-21.699 (7.633)***						
Sunday Rainfall Shock	~	2.652	~	4.737						
Monday Rainfall Shock		(10.100) -1.739		5.341 5.341						
Tuesday Rainfall Shock		(8.873 8.873 (7.500)		(0000) 2.988 2.988						
Wednesday Rainfall Shock		(7.200) (.379		(5.1.30) 2.373						
Thursday Rainfall Shock		(6.903) -7.824		(6.938) -2.234						
Friday Rainfall Shock		(6.790) -2.799		(8.569) -4.121						
# Sat(Rainfall> x mm)		(+00.C)		(601.0)	-0.258 (0.086)***	-0.257 $(0.092)^{***}$	-0.297 (0.087)***	-0.259 (0.091)***	-0.376 (0.094)***	-0.273 (0.092)
# Sun(Rainfall> x mm)					~	0.161	~	0.037	~	-0.026
# Mon(Rainfall> x mm)						(0.065) -0.034		(0.09)		-0.091 0.091
# Tue(Rainfall> x mm)						(0.084) 0.004		(0.075) 0.010		(0.087 0.096
# Wed(Rainfall> x mm)						(0.081) - 0.055		(0.077) -0.041		(0.085 0.036
# Thur Rainfall × x mm)						(0.098)		(0.117)		(0.117
						(0.100)		(0.111)		(0.113
# ŀrı(Kaıntall> x mm)						-0.128 (0.106)		-0.029 (0.111)		-0.186 (0.120
Standard Controls	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Commune Effects	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
\mathbb{R}^2	0.53	0.53	0.53	0.53	0.53	0.53	0.53	0.53	0.53	0.53
Z	1433	1433	1433	1433	1433	1433	1433	1433	1433	1433

Table A.4: Alternative Rainfall Measures

our period of interest minus long-term average rainfall (long-term number of Saturdays with rainfall above 10 mm) normalized by its long-term standard deviation (and similarly for all other weekdays). The long-term averages/standard deviations are for the period January 1984 until September 1990. In regressions 5 to 10 the rainfall thresholds are defined as *t* times the standard deviation of the long-term average plus the long-term average for each sector, where *t* is such that the sector-specific cutoffs match our usual cutoffs on average (i.e. 9 mm, 10 mm, and 11 mm). **# Civilian Perpetrators per Hutu (p.H)** is measured in percent. **Standard Controls** include Hutu population. All regressions are run using weighted least squares (WLS) estimation with Hutu population size as weights. There are **142 communes** in the sample. **Standard errors** are clustered at the commune level. *significant at 10 percent, **significant at 5 percent, ***significant at 1 percent.

Dependent variable:		# Civilia	n Perpetrators,	p.H.	
			Thresholds		
	8 mm	9 mm	10 mm	11 mm	12 mm
	(1)	(2)	(3)	(4)	(5)
# Sat(Rainfall> x mm)	-0.315	-0.351	-0.414	-0.379	-0.374
	$(0.128)^{**}$	$(0.110)^{***}$	$(0.120)^{***}$	$(0.113)^{***}$	$(0.127)^{***}$
x Opposition Leaders	1.202	1.653	1.096	1.063	1.053
	(0.759)	$(0.768)^{**}$	(0.769)	$(0.572)^{*}$	$(0.565)^{*}$
x Tutsi Minority Share	-1.577	-0.903	-0.800	-2.635	-2.046
	(1.581)	(1.197)	(1.407)	$(1.427)^{*}$	$(1.050)^{*}$
Standard Controls	yes	yes	yes	yes	yes
Other Weekday Controls	yes	yes	yes	yes	yes
Commune Effects	yes	yes	yes	yes	yes
\mathbb{R}^2	0.53	0.54	0.53	0.54	0.53
N	1427	1427	1427	1427	1427

 Table A.5: Differential Effects in Opposition Sectors – Interactions (Alternative Thresholds)

to March 1994. All regressions are run using weighted least squares (WLS) estimation with Hutu population size as weights. There are **142 communes** in the sample. **Standard errors** are clustered at the commune level. *significant at 10 percent, **significant at 5 is measured in percent. Opposition Leader is a dummy taking on the value of 1 if the commune in which the sector lies had an Hutu population, average daily rainfall for January 1984 to September 1990 and average daily rainfall for October 1990 to March 1994. Other Weekday Controls include the number of Sun/Mon/Tue/Wed/Thu/Fri with rainfall above *x* mm during the period October 1990. Notes: # Sat(Rainfall>x mm) is the number of Saturdays with rainfall above x mm during the period October 1990 to March 1994 (and similarly for all other weekdays). The thresholds x are given in the column headers. # Civilian Perpetrators per Hutu (p.H) opposition party leader. Tutsi Minority Share is the fraction of Tutsi divided by the fraction of Hutu. Standard Controls include percent, ***significant at 1 percent.

Dependent Variable:	# Civilian Perpetrators, p.H.						
	(1)	(2)	(3)	(4)			
# Sat(Rainfall>10 mm)	$-0.295 \ (0.115)^{**}$	-0.265 (0.123)**	-0.308 (0.113)***	-0.308 (0.108)***			
# Sun(Rainfall>10 mm)	0.045 (0.097)	0.040 (0.105)	0.048 (0.096)	0.043 (0.101)			
# Mon(Rainfall>10 mm)	0.084 (0.102)	0.085 (0.099)	0.044 (0.100)	0.052 (0.101)			
# Tue(Rainfall>10 mm)	0.035 (0.079)	-0.044 (0.087)	0.024 (0.079)	-0.057 (0.088)			
# Wed(Rainfall>10 mm)	0.013 (0.107)	0.043 (0.096)	0.003 (0.105)	-0.025 (0.101)			
# Thu(Rainfall>10 mm)	-0.041 (0.129)	-0.027 (0.131)	-0.025 (0.126)	-0.010 (0.120)			
# Fri(Rainfall>10 mm)	-0.091 (0.097)	-0.091 (0.103)	-0.059 (0.095)	-0.046 (0.089)			
Standard Controls	yes	yes	yes	yes			
Weekday Rainfall, all Transmitters	yes	yes	no	no			
Max. Weekday Rainfall, all Transmitters	no	yes	no	no			
Weekday Rainfall, Closest Transmitter	no	no	yes	yes			
Max. Weekday Rainfall, Closest Transmitter	no	no	no	yes			
Commune Effects	yes	yes	yes	yes			
R ² N	0.54 1433	0.54 1433	0.53 1433	0.54 1433			

 Table A.6: Controlling for Potential Radio Fade

Notes: **# Sat(Rainfall>10 mm)** is the number of Saturdays with rainfall above 10 mm during the period October 1990 to March 1994 (and similarly for all other weekdays). **# Civilian Perpetrators per Hutu (p.H)** is measured in percent. **Standard Controls** include Hutu population, average daily rainfall for January 1984 to September 1990 and average daily rainfall for October 1990 to March 1994. **Weekday Rainfall, all Transmitters** is the number of Saturdays with rainfall along the way between sector and the two RTLM transmitters above 10 mm (and similarly for all other weekdays). **Max. Weekday Rainfall, all Transmitters** is the number of Saturdays with rainfall along the way between sector and the two RTLM transmitters above 10 mm (and similarly for all other weekdays). **Max. Weekday Rainfall, all Transmitters** is the number of Saturdays with maximum rainfall along the way between sector and the two RTLM transmitter and the closest RTLM transmitter above 10 mm (and similarly for all other weekdays). **Max. Weekday Rainfall, Closest Transmitter** is the number of Saturdays with maximum rainfall along the way between sector and closest RTLM transmitter above 10 mm (and similarly for all other weekdays). **Max. Weekday Rainfall, Closest Transmitter** is the number of Saturdays with maximum rainfall along the way between sector and closest RTLM transmitter above 10 mm (and similarly for all other weekdays). **Max. Weekday Rainfall, Closest Transmitter** is the number of Saturdays with maximum rainfall along the way between sector and closest RTLM transmitter above 10 mm (and similarly for all other weekdays). **All** regressions are run using weighted least squares (WLS) estimation with Hutu population size as weights. There are **142 communes** in the sample. **Standard errors** are clustered at the commune level. *significant at 10 percent, **significant at 5 percent, ***significant at 1 percent.

Dependent variable:				# Civilia	in Perpetrators,	p.H.			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
# Sat(Rainfall>10 mm) - Pre-Politi	cal Umuganda	Period							
Period 84-90: whole period	0.017 (0.055)								
Period 84-90: 1. 27 Months	(01000)	-0.199	-0.166	-0.178	-0.168	-0.184	-0.170	-0.201	-0.191
Period 84-90: 2. 27 Months		0.082	0.137	0.132	0.142	0.134	0.128	0.132	0.127
Period 84-90: 3. 27 Months		(0.127) 0.082 (0.168)	(0.125) 0.133 (0.163)	(0.125) 0.129 (0.162)	(0.125) 0.108 (0.163)	(0.128) 0.141 (0.166)	(0.124) 0.132 (0.180)	(0.123) 0.158 (0.160)	(0.121) 0.150 (0.176)
# Sat(Rainfall>10 mm) - Political V	Jmuganda Perio	od							
Period 90-94: whole period			-0.404 (0.102)***						
Period 90-94: 1. 14 Months			(-0.327					
Period 90-94: 2. 14 Months				-0.148 (0.176)					
Period 90-94: 3. 14 Months				-0.583					
Period 90-94: 1. 7 Months				(01100)	-0.093				
Period 90-94: 2. 7 Months					(0.192) -0.548 $(0.256)^{**}$				
Period 90-94: 3. 7 Months					0.100				
Period 90-94: 4. 7 Months					(0.200) -0.447 (0.250)*				
Period 90-94: 5. 7 Months					-0.415				
Period 90-94: 6. 7 Months					(0.206) -0.756 $(0.225)^{***}$				
Period 90-94: 1. 3.5 Months					(0.225)	-0.220	-0.248		
Period 90-94: 2. 3.5 Months						(0.331) -0.088 (0.201)	-0.130		
Period 90-94: 3. 3.5 Months						(0.501) -0.794 (0.621)	(0.522) -0.539 (0.717)		
Period 90-94: 4. 3.5 Months						(0.021) -0.623 $(0.207)^{**}$	(0.717) -0.728 $(0.201)^{**}$		
Period 90-94: 5. 3.5 Months						(0.297) -0.234 (0.400)	(0.301) -0.339 (0.205)		
Period 90-94: 6. 3.5 Months						(0.400) 0.419 (0.434)	0.341		
Period 90-94: 7. 3.5 Months						-0.556	-0.552		
Period 90-94: 8. 3.5 Months						-0.345 (0.260)	-0.418		
Period 90-94: 9. 3.5 Months						-0.276	(0.291) -0.141 (0.218)		
Period 90-94: 10. 3.5 Months						(0.207) -2.054 $(0.602)^{***}$	(0.218) -1.796 $(0.500)^{***}$		
Period 90-94: 11. 3.5 Months						(0.003) -0.841 $(0.260)^{***}$	(0.399) -0.749 $(0.275)^{***}$		
Period 90-94: 12. 3.5 Months						(0.209) -0.552 $(0.317)^*$	(0.273) -0.556 $(0.314)^*$		
Period 90-94: before RTLM						(0.517)	(0.514)	-0.219	-0.236
Period 90-94: after RTLM								(0.114) -0.827 $(0.215)^{***}$	(0.117) -0.777 $(0.205)^{***}$
Standard Controls Other Weekday Controls Additional Controls Commune Effects	yes yes no yes	yes yes no yes	yes yes no yes	yes yes no yes	yes yes no yes	yes yes no yes	yes yes yes yes	yes yes no yes	yes yes yes
R ² N	0.55 1266	0.55 1266	0.56 1266	0.56 1266	0.56 1266	0.56 1266	0.57 1266	0.56 1266	0.57 1266

Table A.7: Mechanism: Effects over Time – Pro-Hutu Sectors

Notes: All of the regressors in this table capture the number of Saturdays with rainfall above 10 mm for different time periods. **Regressors 1** is the number of rainy Saturdays for the period January 1984 to September 1990. **Regressors 2 to 4** are the number of rainy Saturdays for the period January 1984 to September 1990 (split into three time intervals of equal length). **All the other regressors** are from the time period October 1990 to March 1994 (split into various different time intervals of equal length, i.e. 3, 6, 12, and before/after RTLM). **# Civilian Perpetrators per Hutu (p.H)** is measured in percent. **Standard Controls** include Hutu population, average daily rainfall for January 1984 to September 1990 and average daily rainfall for October 1990 to March 1994. **Other Weekday Controls** include the number of Sun/Mon/Tue/Wed/Thu/Fri with rainfall above 10 mm during the period October 1990 to March 1994. **Other Weekday Controls** include the number of Sun/Mon/Tue/Wed/Thu/Fri with rainfall above 10 kgail, Nyanza, the border, the closest main road and the closest main city and town. All regressions are run using weighted least squares (WLS) estimation with Hutu population size as weights. There are **142 communes** in the sample. **Standard errors** are clustered at the commune level. *significant at 1 percent.
Dependent variable:		# Civili	an Perpetrators	s, p.H.	
	(1)	(2)	(3)	(4)	(5)
# Sat(Rainfall> 10 mm) - Pre-Political Umuganda Period					
Period 84-90: whole period	-0.644 (0.614)				
Period 84-90: 1.27 Months	~	-1.435	-1.444	-1.445	-1.606
Period 84-90: 2. 27 Months		(1.257) - 0.051	(1.178) -0.204	(1.003) -0.304	(1.243) 0.198
Period 84-00: 3 27 Months		(0.443) -0 382	(0.604) -0.586	(0.638) -1 148	(0.723) -1 119
		(0.691)	(0.861)	(1.005)	(1.155)
# Sat(Rainfall> 10 mm) - Political Umuganda Period					
Period 90-94: whole period			0.555 (0.952)		
Period 90-94: before RTLM				0.950	0.507
				(1.004)	(0.996)
Period 90-94: after RTLM				-1.472 (1.339)	-0.959 (1.441)
Standard Controls	yes	yes	yes	yes	yes
Other Weekday Controls	yes	yes	yes	yes	yes
Additional Controls	ou	ou	ou	ou	yes
Commune Effects	yes	yes	yes	yes	yes
\mathbb{R}^2	0.36	0.36	0.37	0.38	0.41
Ν	161	161	161	161	161

 Table A.8: Mechanism: Effects over Time – Opposition Sectors

October 1990 to March 1994 (split into various different time intervals of equal length, i.e. before/after RTLM). # Civilian Perpetrators per Hutu (p.H) is measured in percent. Standard Controls include Hutu population, average daily rainfall for January 1984 to September 1990 1 is the number of rainy Saturdays for the period Janary 1984 to September 1990. **Regressors 2 to 4** are the number of rainy Saturdays for squares (WLS) estimation with Hutu population size as weights. There are 142 communes in the sample. Standard errors are clustered at the the period Janary 1984 to September 1990 (split into three time intervals of equal length). All the other regressors are from the time period and average daily rainfall for October 1990 to March 1994. Other Weekday Controls include the number of Sun/Mon/Tue/Wed/Thu/Fri with rainfall above 10 mm during the period October 1990 to March 1994. Additional Controls are sector ruggedness, Hutu population density, distance to Kigali, Nyanza, the border, the closest main road and the closest main city and town. All regressions are run using weighted least commune level. *significant at 10 percent, **significant at 5 percent, ***significant at 1 percent.

Dependent variable:					#	Civilian Perpe	trators, p.H.					
	8	mm Threshold		6	mm Threshold		11	mm Threshold		121	nm Threshold	
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
# Sat(Rainfall> x mm) - Pre-Political Umuganda	a Period											
Period 84-90: whole period												
Period 84-90: 1. 27 Months	-0.197	-0.011	0.520	-0.590	-0.383	-0.077	-0.756	-0.681	-0.533	-0.104	0.536	0.873
Period 84-90: 2. 27 Months	(0.300) -0.445	(0.338) -0.568	(0.531) - 0.432	(0.489) -0.034	(0.433) -0.492	(0.507) - 0.732	(0.626) -0.110	(0000) -0.021	(0.604) 0.172	(1.399) - 0.323	(1.168) -0.315	(1.089) - 0.418
Period 84-90: 3. 27 Months	(0.506) -1.058	(0.508) -1.222	(0.625) -1.253	(0.558) -0.689	(0.558) -1.137	(0.587) -1.726	(0.901) - 0.034	(0.894) -0.210	(0.851) -1.082	(0.608) -0.110	(0.615) - 0.876	(0.569) -0.886
	(0.668)	(0.758)	(0.839)	(0.651)	$(0.615)^{*}$	$(0.935)^{*}$	(0.791)	(0.726)	(1.330)	(0.598)	$(0.492)^{*}$	(0.740)
# Sat(Rainfall> x mm) - Political Umuganda Per	riod											
Period 90-94: whole period	0.613 (0.622)			0.907 (0.683)			0.301 (0.865)			1.186 (0.945)		
Period 90-94: before RTLM		1.027	1.021		1.682	1.754		1.014	1.013		2.340	1.961
Period 90-94: after RTLM		(0.782) -1.087 (0.965)	(20.0) -1.001 (0.916)		$(1.021)^{(0.021)}$ -2.290 $(1.131)^{*}$	(200) -2.837 $(1.174)^{**}$		(0.818) -2.150 $(1.132)^{*}$	$(1.285)^{(0.92)}$		(0.90/) -2.828 (1.689)	(1.001) -3.371 $(1.836)^{*}$
Standard Controls	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Other Weekday Controls	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Additional Controls	ou	no	yes	ou	ou	yes	no	ou	yes	ou	no	yes
Commune Entects	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
\mathbb{R}^2	0.39	0.40	0.43	0.40	0.43	0.45	0.35	0.36	0.40	0.39	0.41	0.44
Ν	161	161	161	161	161	161	161	161	161	161	161	161
Notes: All of the regressors in this table capture th	ne number of S	aturdays with	rainfall above	x mm for diff	stent time perio	ds. The threshe	old x is defined	d in the colum	n header. Regr	cessor 1 is the	number of rair	iy Saturdays
for the period Janary 1984 to September 1990. Re g from the time period October 1990 to March 1994	gressors 2 to 4 4 (split into var	are the numbe ious different	rr of rainy Sati time intervals	urdays for the p of equal lengtl	beriod Janary 19 h. i.e. before/af	<pre>84 to Septemb ter RTLM). # 0</pre>	er 1990 (split Z ivilian Perp	into three time etrators per F	intervals of eq [utu (p.H) is n	qual length). A neasured in pe	II the other re- rcent. Standa	gressors are rd Controls
include Hutu ropulation, average daily rainfall for	January 1984	to September	1990 and aver	age daily rainfa	all for October	1990 to March	1994. Other	Weekday Con	trols include the	he number of S	Sun/Mon/Tue/	Ved/Thu/Fri
main city and town. All regressions are run using	weighted leas	t squares (WL)	S) estimation	with Hutu pop	ulation size as '	veights. There	are 142 com	munes in the s	yanza, ure ooi ample. Standa	ard errors are	clustered at th	le commune
level. *significant at 10 percent, **significant at 5_1	percent, ***sig	gnificant at 1 p	ercent.			1						

 Table A.9: Mechanism: Effects over Time – Opposition Sectors (Alternative Thresholds)

Dependent Variable:	#	Civilian Perp	etrators, p.H.	
	(1)	(2)	(3)	(4)
# Sun-Fri(Rainfall>10 mm)	0.022	0.029		
	(0.054)	(0.059)		
x RTLM Coverage	-0.009	-0.05		
	(0.00)	(0.00)		
# Sun-Fri(Rainfall>10 mm), before RTLM			0.022	0.015
			(0.065)	(0.072)
x RTLM Coverage			-0.003	0.004
			(0.025)	(0.029)
# Sun-Fri(Rainfall>10 mm), after RTLM			0.032	060.0
			(0.113)	(0.128)
x RTLM Coverage			-0.025	-0.032
			(0.071)	(0.086)
Standard Controls	yes	yes	yes	yes
Saturday Weekday Controls	yes	yes	yes	yes
Additional Controls	no	yes	ou	yes
RTLM Controls	yes	yes	yes	yes
Commune Effects	yes	yes	yes	yes
\mathbb{R}^2	0.49	0.50	0.49	0.50
Ν	1057	1057	1057	1057
Notes: # Sun-Fri(Rainfall>10 mm) is the number of weekda; October 1990 to March 1994. # of Sun-Fri(Rainfall>10 mm urdays) with rainfall above 10 mm during the period October the fraction of the sector that receives the RTLM signal. # Civi Controls include Hutu population, average daily rainfall for Ja	/s (except for Sature (), before (after) R' 1990 to June 1993 lian Perpetrators I nuary 1984 to Septe	lays) with rainfal FLM is the numb (July 1993 to Ma er Hutu (p.H) is ember 1990 and a	l above 10 mm du ber of weekdays (urch 1994). RTL , s measured in per- iverage daily rain:	rring the period except for Sat- M Coverage is cent. Standard call for October

1990 to March 1994 (regressions 1 and 2) and the number of Saturdays with rainfall above 10 mm before and after RTLM (regressions 3 and 4). Additional Controls are sector ruggedness, Hutu population density, distance to Kigali, Nyanza, the border, the closest main road and the closest main city and town. RTLM Controls are distance to the closest RTLM transmitter as well as mean and variance

1990 to March 1994. Saturday Weekday Controls are the number of Saturdays with rainfall above 10 mm during the period October

of sector elevation. All regressions are run using weighted least squares (WLS) estimation with Hutu population size as weights. There are **142 communes** in the sample. **Standard errors** are clustered at the commune level. *significant at 10 percent, **significant at 5

percent, ***significant at 1 percent.

Table A.10: Interactions Effects between RTLM and Rainfall on Other Weekdays